The University of San Francisco USF Scholarship: a digital repository @ Gleeson Library | Geschke Center

Doctoral Dissertations

Theses, Dissertations, Capstones and Projects

2009

Demystifying success: critical reflections of African-American California community college students about factors leading to scholastic success during the first year

Robert Lee Snowden

Follow this and additional works at: https://repository.usfca.edu/diss



Part of the Education Commons

Recommended Citation

Snowden, Robert Lee, "Demystifying success: critical reflections of African-American California community college students about factors leading to scholastic success during the first year" (2009). Doctoral Dissertations. 169. https://repository.usfca.edu/diss/169

This Dissertation is brought to you for free and open access by the Theses, Dissertations, Capstones and Projects at USF Scholarship: a digital repository @ Gleeson Library | Geschke Center. It has been accepted for inclusion in Doctoral Dissertations by an authorized administrator of USF Scholarship: a digital repository @ Gleeson Library | Geschke Center. For more information, please contact repository@usfca.edu.

The University of San Francisco

DEMYSTIFYING SUCCESS: CRITICAL REFLECTIONS OF AFRICAN-AMERICAN CALIFORNIA COMMUNITY COLLEGE STUDENTS ABOUT FACTORS LEADING TO SCHOLASTIC SUCCESS DURING THE FIRST YEAR

A Dissertation Presented to
The Faculty of the School of Education
International and Multicultural Education Department

In Partial Fulfillment
of the Requirements for the Degree
Doctor of Education

By Robert L. Snowden Jr. San Francisco, California May 2009

THE UNIVERSITY OF SAN FRANCISCO Dissertation Abstract

Demystifying Success: Critical Reflections of African-American California Community College Sophomores About Factors Leading to Their Academic Success During the First Year

To date a noteworthy amount of research has been conducted on African American student success strategies and the first year experience of college students independently. Researchers including Vincent Tinto (1987), Randi Levitz and Lee Noel (1989), Corinna Ethington (1990), and Shaun Harper (2006), have each explored the issue of student departure at the university level and the success of African American students in higher education separately.

What remained to be explored were the critical reflections and experiences of first year African American California community college students about their persistence, progress towards degree/certificate completion, transfer preparation and their use of student services resources. The purpose of my study was to explore the experiences of African American community college freshmen in an attempt to improve institutional approaches to African American student success. My study utilized analysis of the narrative to explore the critical reflections of six African American community college students about their successful transition from the first-year to the second-year of college. Critical Race Theory was applied as a theoretical lens in reference to understanding the historically unequal circumstances African American students have faced over the history of the American educational system.

The findings from this study indicate that personal connections African American students make with college faculty are critical elements to academic success. Therefore institutions should encourage faculty to make themselves available to students by

becoming active as advisors for student clubs or mentoring students within the instructor's discipline. It's also recommended that African American faculty increase their visibility on campus and utilize the social capital they hold with African American students as a tool for motivation. Lastly, the findings reveal that future African American students would benefit from becoming involved in co-curricular activities early in their academic career and creating connections with second and third year students to further acculturate themselves to the institution.

This dissertation proposal, written under the direction of the candidate's dissertation committee and apported by the members of the committee, has been presented to and accepted by the Faculty of the school of Education in partial fullfillment of the reuirements for the degree of Doctor of Education. The content and research methodologies presented in this work represent the work of the candidate alone.

Robert L. Snowden, Jr.	5/5/09
Candidate	Date
Dissertation Committee	
D. D. (1971)	C 10 100
Dr. Betty Taylor	6/9/09
Chairperson	Date
Dr. Emma Fuentes	5/5/09
,	Date
Dr. Patricia A. Mitchell	5/5/09
	Date

DEDICATION

This work is dedicated to my amazing and supportive family. Mom, without your uncompromising love I would have never made it through my undergraduate experience. You have always been my partner in this journey and this degree is as much yours as it is mine. Aunt Carolyn and Grandma Lelia the two of you have been in my corner from the day I was born, everything you have ever done to help me along has been appreciated and I can't begin to thank you enough. This dissertation is also dedicated to all of my aunts, uncles, cousins and siblings. I know I've been busy for the last decade but now I can finally make those summer barbeques and birthday parties. And to Michele, your genius has inspired me throughout this entire process. You had faith in me, even when I did not; you will always be my rock and thank you helping me cross these burning sands. To everyone not mentioned by name that has been a part of my life throughout this journey thank you and I appreciate you all.

Lastly, this dissertation is dedicated to my father, Dr. Robert L. Snowden. I would like to thank you for giving the inspiration to set and achieve my goals and for blessing me with your infinite wisdom. You have taught me how to be a scholar, a gentlemen and an educator. You have shown me the values of commitment, sacrifice and patience. Most importantly you have taught me the value of family and love. You are the most amazing man I have ever met in my life and I can't thank you enough for being my dad.

ACKNOWLEDGEMENTS

I'd like to acknowledge the dissertation committee, Dr. Betty Taylor (my committee chair), Dr. Emma Fuentes and Dr. Patricia Mitchell. Thank you for your patience and support during the dissertation process and my time at the University of San Francisco. I'd also like to acknowledge Dr. Susan Katz for her support and guidance during my doctoral education.

I'd like to acknowledge the Los Rios Community College district and the participants in this study. Samantha, Iesha, Renee, Serena, Eugene and Q, thank you for your open and honest dialogue about your first year experiences. Lastly, I'd like to acknowledge my mentors Dr. Jamillah Moore, Dr. Janet Hecsh and Dr. Francisco Rodriguez, your unyielding support and your belief in my ability has given me the courage to aim for the stars, thank you all.

TABLE OF CONTENTS

Page
Dedication
List of Tablesviii
CHAPTER I: INTRODUCTION
Statement of the Problem1Background and Need.6Purpose of the Study.8Research Questions.9Theoretical Framework.9Delimitations.14Limitations.14Significance of the Study15Definition of Terms.18Summary.19
CHAPTER II: REVIEW OF THE LITERATURE
Introduction21The First Year Experience21Pre-enrollment Factors22Tinto's Model of Student Departure26Freshman Transition Process27Orientating Students to College Life27Cultural Influence During the First Year of College29Issues Facing African American Students in Higher Education29Summary32CHAPTER III: METHODOLOGY
Introduction35Research Design35Research Setting37Population and Sample38Instrumentation39Data Collection42Data Analysis43Protection of Human Subjects43

Portrait of the Researcher	44
CHAPTER IV: FINDINGS OF THE STUDY	
Introduction	47
Profiles of the Participants	
Findings	
Research Question #1: What Factors Led to the Persistence of	
African American Students During Their First Year at a	
California Community College?	52
Making Personal Connections With College Faculty	52
Research Question # 2: Have College Generated Programs or	
Services Impacted the Success of African American Students	
During the First Year at a California Community College?	61
Research Question #3: What Recommendations Do Current	
African American Students Have For the Institution and Future	
African American First Year California Community College	
Students?	63
Institutional Recommendations	64
Social Integration with the Campus Culture	67
Developing a Personal Relationship With Your Professor	
Summary	69
CHAPTER V: DISSCUSSION, CONCLUSIONS AND RECOMMENDATION	٧S
Introduction	70
Discussion	72
Conclusions	82
Recommendations	83
Recommendations for Future Research	83
Recommendations Professional Practice	84
Summary	
Reflections of the Researcher	86
References	88
APPENDIX A	0/1
IRBPHS Initial Application	
IKBI IIS Illuar Application)3
APPENDIX B	
Los Rios Community College District Research Approval Letter	102
APPENDIX C	103
Student Invitation Letter	
APPENDIX D	105
Letter to College Faculty	

APPENDIX E	107
Release Forms for Tapes and Transcripts	108
APPENDIX F	109
Interview Protocol - Dialogue Questions	110
APPENDIX G	112
Informed Consent Letter	113
APPENDIX H	115
Demographic Questionnaire	116

LIST OF TABLES

Table		Page
1.	Racial and Ethnic Breakdown Percentages for Overall Student Population	4
2.	Age of Students at Enrollment	.4
3.	Gender of Students at Enrollment	.5
4.	Demographics Data of the Student Participants	47

CHAPTER I

INTRODUCTION

Statement of the Problem

The opportunity to earn a proper education is of extreme importance to the advancement of African Americans in the United States (Woodson, 1933). It's unfortunate that the education system over the course of U.S. history has systematically denied equal access in education for African Americans, thus leaving this population historically disadvantaged in respect to access to basic skills and higher education opportunities (Tatum, 2007). Over seventy years ago Woodson (1933) expressed concerns about this phenomenon and the consequences of a dominant group having control over how knowledge is distributed within a society. He argues that the systematic denial of access of equal education between African Americans and Whites has empowered the dominant group in this country with the ability to control the thinking and aspirations of generations of African Americans. He explains,

If you can control a man's thinking you do not have to worry about his action. When you determine what a man shall think you don not have to concern yourself about what he will do. If you make a person feel that he/she is inferior, you do not have to compel him/her to accept an inferior status, he/she will seek for it. If you make a person think that he/she is justly outcast, you don't have to order that person to the back door, that person will go without being told and if there is no back door, the very nature of that person will demand one. (p.iii)

As a result of the continuous unequal and biased education practices in this country, there is a long-standing connection between African American's and a lack of adequate preparedness for higher education (Bennett, 1993).

Minority students in general and African American students specifically are a segment of the California's population that is particularly challenged in matriculating in

to higher education from the states K-12 system. According to the California Department of Education High School Exit Exam Office (2008), African American students earned the lowest passing percentage in both math and English of any student population in the state. Systematically, high percentages of African American students have failed to achieve basic skills competencies in school thus leading to a direct correlation to the dismal numbers of African American students that arrive at colleges and universities each fall ill-equipped for academic success (Hirsch, 2006).

The issues facing the education of students of color in the state of California are vast and without intervention the percentage of African American students that fail to succeed in higher education courses will continue to rise. In the state of California the increasing minority population is eventually going to force a variety of institutions including state government, K-12 education and colleges and universities to address the problems associated with low performing students of color (Harper and Wolley, 2002).

The low percentages of formally educated or vocationally trained African

Americans are having a direct and significant impact on diversity in the workplace and in
the classroom. Based on data in the Community College League of California report
(2008), the California Community College system currently serves the most diverse
population in its history. While the minority population continues to grow the African

American students navigating the community college system are facing overwhelming
social and academic challenges and are consequently failing to achieve proportionate
graduation and transfers rates compared to other populations statewide (California

Community Colleges Chancellor's Office, 2007).

There are a variety of challenges facing institutions that both in theory and practice seek to establish equity between marginalized populations without limiting the opportunities for other groups. As the California community college system continues to grow, steps must be taken to institutionalize methods that consistently address issues associated with educating a diverse student body. These issues include the low performing rates of marginalized student populations and the lack of cultural understanding that can develop from a lack of diversity in classrooms. Despite the fact the California community college system has been a valued part of the collegial matriculation process for millions of students, yet with fewer and fewer African Americans successfully navigating the system, current negative persistence and matriculation trends will undoubtedly continue.

To date a noteworthy amount of research has been conducted on African American student success strategies and the first year experience of college students independently. Researchers such as Vincent Tinto (1987), Randi Levitz and Lee Noel (1989), Corinna Ethington (1990), and Shaun Harper (2006), have each explored the issue of student departure at the university level and the success of African American students in higher education separately. They researchers have found that student engagement and student's feeling of connectedness to the institution are critical to the student's social development and academic success. Though numerous research projects have taken aim at solutions to this issue there continues to be areas of student barriers that should be studied and documented.

What remained to be explored in this area were the critical reflections and experiences of first year African American California community college students about

their persistence, progress towards degree/certificate completion, transfer preparation and their use of student services resources. This study explored the successful strategies that have lead to the persistence of African American students in the California community college system and is designed to serve as a resource for college administrators and faculty who wish to explore methods for affecting the growth and matriculation of African American students on their campuses. As the California community college system advances its efforts in improving the basic skills comprehension and career and technical training this study aligns directly with efforts to erase current trends in education that are perpetuating educational inequality among students of color.

The college chosen for this study is part of the Los Rios Community College

District and is located in Northern California. According to the California Community

Colleges System Office report (2007) the college serves over 18,000 students that reflect
the diversity of a major metropolitan area. Table 1 describes the racial and ethnic student
population at the selected institution, Table 2 describes the age range of the colleges'
student population and Table 3 serves as a description of the gender breakdown of the
campus. In 2005, 14.7 percent of the student body was African American, 22.2 percent
was Asian, 6 percent was Filipino, 31.4 percent identified as white while 14.8 percent
identified as Hispanic. The remaining percentages of the student body were divided up
as other non-white, Pacific Islander, Native American and those who declined to state.
Additionally the students attended the college from all over the region and nearly 60% of
the student population was under the age of 25. The College Research Office Report
(2007) stated that nearly 60% of the student population is female and that the college is
the most diverse among student's ethnicity within the Los Rios District.

Table 1

Racial and Ethnic Breakdown Percentages for Overall Student Population

Racial or Ethnic Background	Overall School Population % (n=18,413)
Asian	22.2
Black African/American	14.7
Filipino	6.0
Hispanic	14.8
White	31.4
Native American	0.9
Pacific Islander	1.7
Other Non-White	2.5
Unknown/Decline to State	5.8

Table 2

Age of Students at Enrollment

Age	Overall School Population % (n=18,413)
Under 18	3.1
18-24	53.8
25-49	36.8
Over 49	6.4
Unknown/Decline to State	0.0

Table 3
Gender of Students at Enrollment

Gender	Overall School Population % (n=18,413)
Female Male Linkmann/Decline to State	58.0 41.0
Unknown/Decline to State	1.0

According to the same report, over 50% of the student population attend exclusively during the day while 25% attend exclusively during the evening. More than 20% of the students attend both day and evening, moreover nearly 30% of the student's reported that their household income is below the poverty level while nearly 70% of the student population is working either full or at least part time. In addition, more than 50% of the student population stated that their educational goal is to transfer to a four-year university and about 20% of students have an educational goal of obtaining an associates degree or certificate.

Background and Need for the Study

African Americans have aspired and achieved academically in spite of tremendous obstacles, yet have historically ranked lower than Whites in academic achievement. This disparity between the earned grade point averages and standardized test scores of African Americans or any other minority population and Whites in the United States is defined as the achievement gap (Hale, 2001). Though as a population African Americans have made significant gains in academics, gains have been made by other minority populations and over time these other groups have been successful in

improving their standing academically in comparison to Whites. African Americans are still marred by a significant disparity between their academic success and the academic accomplishments of Whites and other racial groups (Tatum, 2007). It's critical that educational institutions recognize the inequalities exists between African Americans and other student populations and make deliberate efforts to derail the historical legacy of under achievement and disproportionate success rates.

For many African American students' choosing to attend college is a monumental decision that can lead to substantive change in their current and future lifestyles. If these students have been victims of the inequalities perpetrated throughout k-12 systems then without intervention they are more likely to struggle academically and socially in institutions of higher education. This dissertation examines specifically the experiences of African American students at a Northern California community college that have successfully completed their first year of courses in an effort to gain insight and provide new strategies in promoting African American student success.

Though the college selected for this dissertation has consistently provided excellent service to the community it still has identified areas for improvement. The California Community Colleges System Office (2007) reports that the college is facing challenges in the area of student academic success and matriculation. The report also highlighted that the college is specifically lacking in support mechanisms that would better serve a multicultural population of first year students from urban areas within the area of basic skills. For the purpose of this dissertation basic learning skills has been defined as competency in college level English and college level mathematics courses. Do to the fact basic skills mastery is a requirement for all degree programs at the college

and are a cornerstone of the learning process it is imperative that the college investigate applicable solutions in improving the overall success of a substantial percentage of it's student population.

This dissertation's value is in its focus on a segment of the community college student body that has been identified by this institution and the state of California as suffering from poor matriculation and persistence rates (California Community Colleges System Office, 2007). The data gathered by the researcher from the reflections of these students is designed to lead to overall efforts to enhance the retention and matriculation of African American college freshman by informing faculty, staff and administrators of student perspectives in relation to their own education. Its this researchers intent to provide data to support innovative approaches to institutionalizing improvements for the Academic progress of African American students in higher education.

Purpose of the Study

The purpose of this dissertation was to explore through the reflections of students the factors that lead to African American community college freshman persistence. This dissertation: 1) examined college programs or services that were available to or utilized by the student during the first year and, 2) explored the recommendations of African American college sophomores for incoming African American first semester students. For the purpose of this dissertation student persistence was defined as continued attendance and unit completion from the semester of first enrollment through two consecutive semesters as a full time student according to number of units attempted and completed. Limiting the dissertation to a northern California urban/suburban community college expanded upon previous research on freshman success (Tinto, 1993), African

American college student success, and African American student in the California community college system.

Research Questions

The following questions guided this study:

- 1. What factors led to the persistence of African American students during their first year at a California community college?
- 2. Have college generated programs or services impacted the success of African American students during the first year at a California community college?
- 3. What recommendations do current African American students have for the institution and future African American first year California community college students?

Theoretical Framework

The theoretical framework applied to the reflections of African American first year students in this dissertation was Critical Race Theory. A Critical Race Theory framework underscores the point that race [still] matters when researching issues in the field of education (West, 1992). Critical Race Theory therefore is a fitting context to examine the experiences of African American students as they continue to struggle with matriculation and academic proficiency within the California's Community College system. Early writings on the topic of Critical Race Theory are linked to the writings of Derrick Bell (1993) and Alan Freeman in the mid-1970s (Delgado, 1995). Other writers have contributed to the body of knowledge relating to Critical Race Theory including Kimberlé W. Crenshaw (1995) and E. Nathaniel Gates (1997).

Critical Race Theory was not the first adaption of Critical Theory; Critical Race
Theory was born from a movement called Critical Legal Studies. Critical Legal Studies
is a progressive legal movement that challenges modern legal practices and proposes that
laws address people through social and cultural contexts (Gordon, 1999). Critical Legal
Studies scholars are critical of how mainstream legal ideology and its portrayal of US
society fails to include racism in its critique, thus establishing the grounds for the
development of a restructured method of examining society, Critical Race Theory
(Ladson-Billings, 1998).

According to Delgado (1995,p.xiv), racism is "normal, not aberrant, in American society", because it is so engrained in our history, racism and the social construction of race are natural elements. Yet Critical Race Theory differs from critical legal studies in its ability to incorporate the voices of theorists and the subjects in a way that fosters a humanistic element to the research being conducted. The experiences of both the researcher and the subject create a context where a new analytical perspective about race is established. Not all theorist or scholars of Critical Race Theory hold the same views or apply the theory in the same method. Critical Race Theory has begun to gain exposure in the field of education particularly due to its use of the narrative in examining issues related to schools and learning, early Critical Race Theory scholars such as Patricia Williams (1991) and Derrick Bell (1987, 1992) added to the body of work associated with Critical Race Theory incorporating narrative analysis as a growing trend in qualitative research methods.

Critical Race Theory allows me as the researcher to analyze the social construction of race as a common experience of people of color in the United States.

Race and races are merely the products of ideas and relationships between human beings (Delgado & Stefancic 2001). The social construction of race is based on categories that are invented by society and can be manipulated when beneficial to the dominant culture. In addition to the social construction of race, Critical Race Theory addresses the concept of material determinism. Delgado and Stefancic state that, "Because racism advances the interests of both white elites (materially) and working class people (psychically), large segments of society have little incentive to eradicate it." These exterior factors are extremely important when considering the social and psychological factors that are involved in the process of the higher education for African American students.

The United States has fostered over two hundred years of socially constructed racism towards African American people. That history of racism has had horrific effects on the academic success of students of color throughout this nations history (Green, 1989). Throughout the history of the United States, the federal government has been methodical in its attempt to remove from record numerous laws and practices based in Jim Crow legislation that continue to handicap black school children today. Sadly the roots of past racism are still prevalent in multiple aspects of North American society today (Delgado, 1995). Public education in the United States is the foundation upon which higher education rests and higher education has historically been a means to advance as a people within a society that provides more opportunity to people with formal training and education. Due to socioeconomic limitations, staggering illiteracy rates, and other socially constructed harms, many marginalized groups in the United States are caught within a crippling cycle of poverty and despair. As a result these

marginalized populations are not able to create the social capital needed to make advancements in the United States, due to the lack of a formal education foundation.

The African American community in the United States has measured below average in academic achievement for generations. Across state borders, time zones and socioeconomic status, black students in the United States are not reaching the same levels of academic success as non-black students, nor are they granted many of the same opportunities for academic enrichment as students that are non-black (Daniel and Armida 2004). This gap in academic success is fueled by decades of poor economic conditions, and partially through an educational system that still has a limited desire in teaching students from diverse cultural backgrounds. Due to the staggering deficit that black students are facing because of ineffective schools; alternative strategies are being explored in an attempt to raise the academic success of these disadvantaged students.

Critical Race Theory is being implemented in some settings to help change how students, teachers, administrators and communities challenge themselves to make positive changes in the education of black students (Stovall 2004). I think that because of the intentional and unintentional racism present in schools across the United States, African American students and their communities have over time become desensitized to the importance of a quality education. This is particularly devastating to the African American community because, without the basic knowledge and social conditioning received while receiving a formal education; black people in the United States are being twice defeated. Additionally, by not receiving effective social and academic schooling, African American people are limited in their ability to break the lasting effects that slavery has had on this population (Young, Wright et al. 2005).

Stories and the voices of real people lend validity to issues in education that are complicated by the occasionally subtle and sometimes blatant undercurrent of race and racism in the United States. Critical Race researchers employ narratives as a means of establishing cohesion within a particular race group, and apply these stories as a method of demystifying the assumptions of the dominant group (Delgado & Stefancic, 2001). Critical Race Theory allows the research participants to self identify their own reality, while providing a applicable framework to conduct analysis that can be used locally or generalized to larger segments of the population because of the fluid nature of the subject of race. The experiences of people allow critical race theorists to examine individuals within the context of their own socially constructed realities, rather than imposing a false or manufactured framework to apply their data. This serves as a powerful method of research and can be utilized as a method for exposing the dominant culture to the true nature of their actions on the group being studied.

Critical Race Theory also serves as a method of further understanding the social and psychological state of African American students in higher education. For many African American students, going to college creates a separation from community and family. This population of students is extremely sensitive to outside distractions and is continuously struggling to maintain an authentic connection to all matters that are genuine to their personal history (Hefner, 2004). Race, class and poverty each play parts in the ideological distribution of the material benefits afforded to people in society.

Moreover this distribution of benefits is directly linked to the opportunities and quality of education afforded to African American students especially in urban areas.

Without recognition of the multifaceted means in which race dictates access in the United

States a study calling attention to African American students would be incomplete.

Critical Race Theory as a framework allows for the researcher to include the population being studied, in the critical design of interventions set to assist them attain their scholastic goals. Moreover it allows for the consideration of authentic data in work committed to addressing the distinctive problems associated in society in education as identified by African American students.

Delimitations

This dissertation was conducted at one California community college. The students who were used as participants in the dissertation may differ from students at other community colleges or students of four-year universities. The sample was limited to students that were first time freshman during the fall semester of 2007, therefore students who entered prior to that semester or after that semester may have different experiences due to the fluctuating nature of campus culture and student populations. Additionally, the researcher attempted to recruit ten students for the dissertation and after they were initially self-selected four were disqualified from participating by not meeting predetermined criteria, thus allowing only six participants to contribute as participants in this dissertation.

Limitations

The nature of qualitative research is to gain a rich and personal account of the individual experiences of participants (Creswell, 2005). Therefore the personal experiences of these students may not be applicable to a larger group of first year students regionally or nationally. Moreover, students at the community college level may have different experiences than those students at a private or public four-year institution.

Additionally this community college serves a diverse population of over 18,000 students in a major metropolitan area and may not be relevant to rural or smaller colleges. Because only six students were selected for this dissertation from a first year population the results may not be accurately generalized for the entire freshman class including those students who are classified as part-time. Moreover it should be noted that the experiences of this student population could be affected by the college's efforts to assist first year students through district and statewide initiatives.

The researcher's position as an African American faculty member at the college used in this dissertation may have influenced the responses of the participants as they relate to reflecting candidly about college faculty members or administrators. Four of the six participants had previous co-curricular or social contact with the researcher before this dissertation. Therefore, the researcher acknowledges that the previous level of contact may have created an environment of familiarity for participants when responding to the interview questions.

Significance of the Study

This dissertation examined the experiences of African American students at a California Community College who have successfully completed their first year of academic courses. This dissertation is significant because it focused on a segment of the community college student body that has been identified by the college as suffering from poor matriculation and persistence rates (California Community Colleges System Office, 2007). This researcher believes that the information gathered from the reflections of successful African American students' can lead to the overall enhancement of efforts to retain and support African American college freshmen. This research will serve the

campus by informing administrators of student perspectives in relation to their own education and subsequently motivating administrators and faculty to consider reasonable improvements to the current college infrastructure.

Though community colleges can serve as gateways to higher education, they can work to the benefit of four-year institutions by providing alternative options for students in times of over enrollment (Dowd, 2007). For many undergraduate students seeking a bachelor of arts degree, transferring from a community colleges presents an option for completing a their entire lower division course work before they apply and are admitted to the four-year institution. Despite a growing number of students enrolling in community colleges and the systems ability to assist in transfer to four-year institutions.

Current research supports the perception that the United States is facing a crisis in regards to the scholastic preparedness of African American students (Hefner, 2004).

African American students specifically in the state of California continue to struggle in achieving their transfer goals and improving their rates of matriculation. Therefore it becomes critical that community colleges take a proactive role in supporting this marginalized population in the attainment of their educational goals.

An analysis of the narrative qualitative research approach allowed the students that were a part of this dissertation and opportunity to safely dialogue about their experiences in regards to successes and failures. Additionally this method of research afforded an opportunity for students to act as agents for change at their own institution.

It's my hope that this dissertation will eventually offer valuable recommendations on how the institution should provide programs and services to improve the first year experience specifically for African American students. Additionally this dissertation

showcases the practices of successful African American college students and focuses on proactive methods for producing a greater number of well-adjusted and academically competent African American students.

Definition of Terms

For the purpose of this dissertation, the following terms were defined as follows: *African American*: African American refers to people having origins in any of the Black racial groups of Africa. It includes people who indicated their race or races as "Black, African Am., or Negro" or wrote in entries such as African American, Afro American, Nigerian, or Haitian. (U.S. Census Bureau, 2000)

California community colleges: Higher education system that is comprised of 72 districts and 110 colleges. Community colleges provide vocational training, basic academic skills education and preparation of students for transfer to four-year institutions in addition to offering opportunities for lifelong learning. (California Community Colleges Chancellor's Office, 2007).

Campus services: Campus services refer to support services available to assist students with career decision-making, problem resolution, and goal achievement. (Cosumnes River College Research Office Report, 2007)

Critical Race Theory (CRT): Examines the effects racial prejudice, racial subordination and discrimination within educational institutions in the United States. (Bell, 1993; Delgado, 1995; West, 1992)

Culture: Culture refers to the shared experiences, symbols, values and beliefs of a particular population based on a wide variety of factors. (Banks & Banks, 2003)

Persistence: Persistence is defined as a student enrolling in a particular fall semester who enrolls in the subsequent spring semester. (Cosumnes River College Research Office Report, 2007)

White: White refers to a person having origins in any of the original peoples of Europe, the Middle East, or North Africa. It includes people who indicate their race as "White" or report entries such as Irish, German, Italian, Lebanese, Near Easterner, Arab, or Polish. (U.S. Census Bureau, 2000)

Summary

The motivation for this dissertation was a deep concern for the current rates of matriculation and low academic performance of African American community college students. For those African American first time freshman entering the California community college system, much can be learned from the experiences of African American sophomore students who have successfully navigated through the first year of college. This dissertation is organized into five chapters. Chapter I is comprised of background information supporting the overall need for this dissertation, the statement of the problem, research questions and theoretical framework.

The dissertation's limitations and delimitations are also discussed in chapter I. Chapter II contains a review of significant literature and establishes a theoretical foundation for comprehending the dissertation's findings. Chapter III contains an explanation of the methodology selected for the dissertation, a description of the participants, questions that guided the dissertation, data collection and analysis procedures and the description of how human subjects were protected. Chapter IV contains the dissertation's findings, the profiles of the students and an analysis of their reflections. Lastly, Chapter V includes the recommendations for further research, a discussion of the results, conclusions drawn from the reflections of the students and the final recommendations based on the completed dissertation.

This dissertation analysis the perceptions of African American sophomores that have reflected about some of the critical elements that lead to their academic success during their first year as college students. It is the intention of this dissertation that the analysis of these reflections be used by future students, faculty and college administrators in an effort to create a strategic plan for strengthening the colleges ability to adequately support at risk student populations. By participating in the development strategies for increasing the academic success of African American students the entire campus community can take ownership of the accomplishments and failures of this group, therein working to create an environment of shared accountability and commitment to the overall success of African American students.

CHAPTER II

REVIEW OF THE LITERATURE

Introduction

The literature review is designed to lead to a clear understanding of factors relevant to the first year African American student experience. The review explores the first year experience for college students, retention of first year college students, factors that lead to first year academic success, student expectations of the institution and cultural aspects that can influence the learning experience of African American students.

The first section examines the first year experience, specifically targeting preenrollment factors influencing first year college students and Tinto's model of student
departure. The second section investigates the freshman transition process and relevant
research on first year programs designed to orientate students to college life. The third
section describes cultural elements in the education experience of African American
students. These areas of study are paramount in determining best practices for college
faculty and staff in understanding and supporting African American freshman persistence
and success. The studies reviewed in this section provide a theoretical context for how
academic preparation; family support, culture and college environment can be applied to
the population of interest in this dissertation.

First Year Experience

Every year, first time California community college students face unique and varying challenges to their academic and social growth. When compared to previous undergraduates these students are more socially tolerant; more goal oriented and appear to have a greater focus on career and monetary success (Cawthon and Miller, 2003).

Today's students are more technologically advanced; more socially networked and have a better understanding of cultural awareness than any generation before them. These factors make this population different from previous groups and many institutions are in a state of transition in order to meet the demands of this new demographic. This dissertation is an investigation into the factors that influence the academic success and persistence of first year African American students at California community colleges.

Pre-enrollment Factors

This section illustrates the literature associated with the first-year or freshman experience as it pertains to student's needs and perceptions. As an instructor at a metropolitan community college within the state of California I can attest to the significance of pre-enrollment factors that are evidenced within much of the literature associated with the college experience. Roberts (1995) stated that there are four types of adjustments that first year college students must face: academic, social, emotional and general. Roberts work supports the notion that students that are from stronger family units and had access to greater resources performed better during the first year of college, while those students who did not come from strong family's or had limited resources performed at lower levels academically.

A contributing factor to the low performance of African American students in college is the poor job K-12 systems are doing in educating this population. Though multiple variable can be attributed to student success before college the disproportionately high numbers African American students are facing obstacles to learning including public safety, low expectations from teachers and inconsistent disciplinary treatment (Anonymous, 2008). Moreover, The lack of cultural

understanding by school administrators and teachers can create obstacles for creating an environment for African American students to thrive academically (Perkins, 2008).

Students come to college with a variety of life experiences that make each student interact with the institution in a unique and personal way. Today's traditional freshman student is different from any previous generation of college student because of his/her experiences with technology, culture, and personal relationships. This generation of first time college students is living within a country that is more diverse than it has ever been and is connected seamlessly with the world through the advantages of technology and access.

Although this generation of college students benefits from social, political and societal changes long standing obstacles such as race and racism still stand as barriers to academic success in college (Harper, 2006). The inclusion of race in the exploration of issues facing African American students is applicable due to the complex nature of race within the United States. Race is a part of the African American college student experience because race is a part of the everyday lives of African American students. Therefore in order to determine authentic solutions to the problems facing African American college students it is imperative to include the notion that even in the 21st century race still matters (West, 1992).

As technology becomes increasingly efficient and customizable it offers new opportunities for students to incorporate it into their educational experiences and preferences. The internet, CD's, television and personal computers have forever changed the tools used by students to gather and process information (Shneiderman, 2003).

Therefore institutions must be aware that this current generation of college students will

inherently demand more technologically from their instructors and campus than ever before. This provides an innovative opportunity for colleges to connect with its first year student population in a manner that is generationally appropriate.

The experiences of students before they arrive at college can affect their ability to succeed academically and adapt to college life. In regards to predictions of students success, studies have shown that a student's previous academic record, standardized test scores, socioeconomic status, race and gender can have an impact on their adaption to an institution of higher learning (Astin, 1993). The research suggests that students come to college with elements such as standardized test scores and high school grade point average's that can often serve as predictors of their academic success and overall satisfaction with the institution (Astin, 1993, 1997; Johnson, 2002; Ward-Roof, 2003; Wolcott, 2006). In addition to the individual elements that students bring to college, today's students are more focused on correlations between degree completion and career advancement (Cawthorn and Miller, 2003).

A common theme in the research of pre-enrollment factors contributing to student success is the support system in place for the student by the family unit. The family's support or lack thereof can be directly related to student persistence during the first year (Tinto, 1993). The transition to college can be difficult for both the students and the family unit therefore when the time comes for students to attend college their emotional state can be influenced by how much their parents support them. Institutions that promote open channels of communication with the parents or provide opportunities for families and students to become orientated to current college expectations provide a stronger support system for their students (Austin, 2003). This communication can

manifest in the form of written letters, materials about the college or orientation programs that are designed for both the student and the parent (Austin, 2003). Direct communication between the institution and the parents can help alleviate questions and concerns about the college's programs and services for both the student and parents.

Within the research on pre-enrollment factors the most significant area of support is in regards to the economic stability of the student and their ability to finance their education. The literature suggests that when parents are able to provide financial support, students are not as worried about financing their own education. Therefore when being financially supported, students are less likely to worry about their economic situation and are less likely to seek employment in order to pay for tuition and supplies (Cawthorn & Miller, 2003).

Lastly the theme of technology and how the students of today are more technologically articulate than those from the last generation of college students was present in much of the research exploring this topic. Factors including access to personal computers and the growth of the internet have provided students with daily interaction with technology in areas of communication and the completion of both simple and complex tasks. Students are currently seeking a high level of technological sophistication from the colleges they choose to attend as they are looking for increased opportunities to integrate education into their lives (Cawthorn & Miller, 2003). This theme is increasingly relevant as colleges seek to create more opportunities for learning and retention through technology.

Tinto's Model of Student Departure

Tinto (1975,1993) was responsible for developing a highly respected model for the means of student persistence and departure. Tinto's work builds on past models of student departure from college in the fields of social theory, organizational sociology and psychology. His work added to the body of knowledge in this field by further expanding on the complexities associated with the college experience. Tinto's work expressly took into account the role of the institution in the matriculation and withdrawal process of a student and refocused the approach to matriculation away from the individual successes or failures of the student. Tinto's work explored the studies of anthropologist Arnold Van Gennep, and authors Emile Durkheim and William Spady. Van Gennep (1960) studied the establishment of membership in traditional societies. This process was later described in his writings as "rites of passage"; these stages included separation, transition, isolation and incorporation. The rites of passage work explored by Van Gennep is conceptually equivalent to a student's persistence and adaption to the world of higher education on a college campus (Tinto, 1993).

Tinto applies the work of Durkheim (1951) and Spady (1970) to his theory of student departure. Durkheim's work argued that to prevent an individuals suicide that social restructuring was required promoting a complete integration into society of the individual. Tinto applied this notion to student departure by relating it to a student's environment including, his/her community as a factor that can have a profound impact on their college experience. Tinto (1993) asserts that the college experience for a student is comprised of both a social and academic component. Each component includes its own complexities, yet both work in tandem in establishing the complete college experience.

Within the model of student departure Tinto (1993) illustrates that two dispositions are responsible for how students react behaviorally to certain situations, these dispositions were identified as goals and commitments. Tinto defines goals by referring to the student's intentions, while commitments refer to that student's compliance in working towards their predetermined goals. Tinto's model suggests that the departures that students make from colleges and universities are stimulated from interactions and experiences with social and academic systems. How the students experiences or interacts with these systems as established by the institution will have a direct impact on that student's completion of established goals. Simply stated negative experiences will deteriorate a students commitment to the institution while positive experiences will solidify that students commitment to the attainment of goals and loyalty to the institution. Berger and Milem's (1999), study builds on Tinto's work and suggests that a student's perception of support from the institution will have an effect on the student's rate of persistence. Therefore if an institution can positively impact the communal and scholastic components of a students experience during the first year, then that student will have a much greater opportunity to succeed both socially and academically.

Freshman Transition Process

Orienting Students to College Life

The beginning of a college career is often viewed as a traditional separation from family and required educational exercises to a higher level of chosen study and social independence. Austin (2003) suggests that institutions should incorporate into preenrollment activities or communication a means to connect the family to the new college

experience. Thus creating a tangible link between the students past experiences and future directions. The transition to college can affect students in a variety of ways including personal health, mental health, social adjustment and academic preparedness (Austin, 2003).

Due to the sheer number of options afforded to students once they have completed high school, colleges and universities must now appeal to multiple interests of the student body or enrollment could cease or decline. Strumpf and Sharer (1993) acknowledged that students are now expecting services from institutions that will assist them in the eventual achievement of their scholastic goals. These expectations can have a direct impact on how the student feels about their college experience from the very beginning and can serve as potential reasons for student departure. Therefore it is extremely important for colleges to incorporate an element of customer services into pre-enrollment or orientation efforts in an attempt to meet the expectations of this new generation of students (Gardner & Hansen, 1993).

Research has shown that students who are able to participate in a first year orientation program or first year experience courses typically perform at higher levels than those who do not (Austin, 2003). Additionally orientation programs for parents have been established in an effort to better condition families for the realities of college life and online or virtual orientations are an option provided by some California community colleges. Based partially on research and on best practices shared from colleagues, institutions across the state are using these varied approaches to strengthen the chances of first year student success.

Research has stated that when students progressed at a higher rate and were more successful during their first year when they were involved in campus activities and within the learning process (Astin, 1984). As research continues in the areas of presentence and retention Astin's work has been used in other studies (eg. Berger & Milem, 1999; Grosset, 1991; Kuh, 2003; Kuh, Shuh, Whitt & associates, 1991; Tinto, 1993). Astin (1996) reexamined his previous study on student involvement and through a review of data concluded that student involvement continues to have influence in student's persistence and attitudes towards the institution. The involvement of students during their first year has remained a constant theme in the research and suggests institutions that create opportunities for early involvement can have an influence in perception and persistence.

Cultural Influence During the First Year of College Issues Facing African American Students Enrolled Higher Education

The diversity of today's community college student population reflects their aptitude to be tolerant of cultural differences and varied educational goals. The Community College League of California, (2008), states that the California Community Colleges (CCC) serves 63.7 out of every 1,000 members of the states population. Of that number 24.6% would be considered traditional first year students, 17-19 years of age and attending higher education for the first time. In terms of gender in 2006, 55.5% of the CCC population was female while 45.5% of the population was made up of males (Community College League of California, 2008). Among this population of students 7.2% self selected as African American, identifying them as the minority group with the third highest percentage of students within the system. Though African Americans are a

significant segment of the minority student population within the community college system, they often face unique outlying factors that can provide challenges to their overall academic success.

The literature based on student success in higher education supports the notion of parental involvement in a student's life and the student's home environment play significant roles in college preparation and projected future academic success. Moreover, it has been found that the decision to attend college and the process of applying to college has an emotional dimension to it and therefore associates a value in education within the family structure (Austin, 2003). Family input is a critical element to the overall achievement of a college student and in the case of the African American family unit; this element can be a crucial support mechanism or very real deterrent to seeking higher education. The family unit can also be expanded to include non-relatives that have influence in the student's life include friends, mentors and members of the community.

The close structure of many African American communities allows for a greater number of influential factors in the student's decision-making process. Whether some African American student's attend or remain in college can be affiliated with the influence of that student's community and their peers. This can be further complicated as many students of color are forced to successfully navigate the two worlds of home and school (Jalomo, 1995). Rendon, Jalomo, and Nora (2000) argue that institutions should work to assist students with this navigation between both worlds especially when diversity and varied cultural perspectives are listed as part of the overall purpose, mission and values of the institution.

In addition to an African American student's family unit, the student's K-12 experience and preparedness for higher education serves as an instrumental piece in predicting that student's success in college. Astin (1997) found students that had high standardized test scores and high grade point averages signifying their mastery of secondary education, typically completed college while those with low standardized test scores and low grade point averages exhibited lower percentages of matriculation and completion. Furthermore it is important to consider the considerable measure of academic disparity that African American students have faced by the ineptness of some teachers to motivate students of color.

A history of poor cultural understanding, improper attitudes and stereotyping toward African American students by society has served to substantially place African American students at a disadvantage in regards to equal access in education. In order to improve the education of African-American student's, colleges and universities must excel in preparing teachers and administrators to have an elevated level of authentic knowledge of African American culture; a deeper understanding of the impact African American culture has on behavior, learning styles, and a personal connection to the institution; and a genuine appreciation for the valuable gamut of experiences African-American students bring to institutions (Ford, Obiakor, & Patton, 1995). Creating and maintaining a minority undergraduate population at institutions of higher learning serves in ensuring equity for minority citizens and maintains a diversified learning environment for all (Green, 1989).

Research on the persistence of African American students in higher education suggests that the issue is complicated and multifaceted. Scholars argue varied reasons for

why this particular population continues to face challenges. Crosson (1988) suggests that the overwhelming nature of hostel racial climates that African American students must face in colleges directly impacts their academic achievement. Blackwell (1983) suggests that African American students more than any other population are less prepared academically to succeed in college therefore leading to poor rates or persistence and completion. Ottinger (1991) sites the poor socioeconomic situation of many African American is critical when considering persistence and ability to comprehend material.

As research continues with a concentration on African American student persistence these studies stand as support for the theory that African American students face a varied and complex set of circumstances when entering higher education. An understanding of this situation is critical if institutions are committed to generating transformative practices in a directed effort to serve this population of students.

Summary

In summary, this review of the literature shows research on the first year experience with a direct emphasis on pre-enrollment factors that contribute to student development, Tinto's foundational model of student departure, the freshman transition process and cultural elements in the experiences of African American students. The study of the first year experience in colleges and universities is complex and varied. Though this review highlights years of comprehensive research, areas for expansion on previous data remain clear.

The California community college system faces a unique and challenging future.

As the system continues to grow more and more students will be looking towards these institutions to complete their lower division coursework or to earn a degree in hopes of

attaining a place of value in the workforce. Consequently it will be increasingly vital for community colleges in the state of California to develop methods to meet the challenges of the new generation of an increasingly multicultural, socially networked, academically deficient, and technologically advanced first year student population entering their institutions. Astin's (1997) research helps frame the theory that direct involvement in the campus will strengthen persistence and success. This theme is present throughout the research about the first year experiences and serves as a constant when evaluating potential interventions for student success. Other factors explored in the literature on the first year experience include the significance of perceived institutional support by the students during their first year and opportunities for co-curricular participation.

The African American student population in higher education will continue to face varied challenges as outlined in the literature. At the forefront of the challenges for community colleges is developing strategies for how these institutions will service students who are not adequately prepared for higher education. Additionally how will African American first year students perform while facing a wide-ranging cadre of obstacles that have historically limited the scholarly progression of this population.

It's the responsibility of the California Community College system to understand the social and historical challenges facing this population and develop a plan of action to assist in the nurturing and development of these young minds. The institutions that make up the network of CCC's will be the first experience in higher education for much of the states African American population and for that reason serves a complicated roll as an institution for advancement to the university system and as foundation for improving basic academic skills. This review was constructed to illustrate the framework necessary

in adding to the current body of knowledge regard first year student success and sets the context for future research on metropolitan community college student success.

CHAPTER III

METHODOLOGY

Introduction

The purpose of this dissertation was to explore the experiences of African American community college freshmen in an attempt to improve institutional approaches to African American student success. Limiting the dissertation to a northern California urban/suburban community college expands upon previous research on freshman success at the community college level. In this dissertation, participants were asked to reflect on how the college served their needs as learners. Participants also reflect on the significant elements of college wide efforts to assist them in persisting during their first year of higher education. By identifying the factors that lead to student persistence this research may serve as an institutional resource for initiating dialogue and action among college administrators seeking to improve current institutional methods for student success.

Research Design

This dissertation used a qualitative research design that utilized analysis of the narrative (Riessman, 1993) as a method of data interpretation, allowing this researcher to extract themes from the stories of African American students who have successfully matriculated through their first year of college. Narrative research allows the researcher to collect data through the telling of the human experience by participants in the study (Elliott, 2005). This research method is used when the stories of individuals can provide unique and personal insight into a particular issue being studied. In analysis of the narrative the researcher works to collect reflections of past events through interviews or

observations and manufactures them into themes to create understanding of a larger question. Key characteristics of narrative research include inclusion of context to the telling of the story, a focus on the experience of the individual and an emphasis on the construction of personal accounts based on information gathered through interviews (Creswell, 2005). Utilizing a qualitative research approach in this dissertation allows the students to make a personal connection with the research process and eventually with the findings of the study.

By implementing this type of qualitative research the students become part of the possible solution and work towards making a positive change for future first semester students. Additionally this research method can empower students' in making change in an effort to improve the understanding of the institution in regards to the academic and social obstacles for African American students during their first year. Utilizing with analysis of the narrative, this researcher plans to identify common themes of success within the student's stories. Once these themes are identified they can serve as a flexible and detailed method for collecting and analyzing information (Bryant, 2006).

Analysis of the narrative serves as an excellent tool for evaluating transcripts of first-person stories or experiences. These first-person accounts are what this researcher plans to use to encourage dialogue between administrators, faculty, and students in developing meaningful campus resources to improve the academic success of African American students.

Research Setting

This dissertation was completed at one Northern California community college. The institution is a part of the Los Rios Community College District and is located in the greater Sacramento metropolitan area. The college opened in 1970, with nearly 1,800 day and 350 evening students (CRC, 2009). Facilities included the science building, automotive and technology complex, women's physical education building, pool, and the library building. Later the college expanded to include, the Business/Social Science classroom building, computer laboratories, physical education buildings and a state of the art center for performing arts (CRC, 2009). Additionally the campus consists of 12 permanent buildings, including up to date fiber optics and satellite transmission resources for the best possible in information/communication services. Lastly, the cafeteria and college center encompass facilities for both the instructional foodservice program and student/staff needs.

The institutions average length of service for tenured faculty is 11 years additionally the longest tenured faculty member has been at the institution for 36 years. According to Los Rios office of human resources report (2006) of the 169 tenured or tenure track faculty at the college the ethnic makeup included 63% white faculty, 11.5% Hispanic faculty, 11% Asian faculty, and 10% African American faculty. The institution and the Los Rios district have recognized that the ethnic composition of the faculty at this campus is not truly reflective the ethnic composition of the student population and strives to promote more diversity within its full time tenure track instructors.

The college offers a broad option of more than 60 occupational degree programs, and a comprehensive curriculum of transfer courses. The institution takes pride in it's

diverse student population with over 50 percent of its' student body representing a variety of ethnic and international backgrounds. The college has constructed its own identity within the second largest community college district in the state of California. Yet still remains small enough to offer personalized attention and services, and large enough to provide extensive postsecondary options to the Northern California community it serves.

Population and Sample

This researcher worked with six African American students in the sharing of reflections about their experiences as first year community college students. The students selected for this dissertation became both subjects and active participants in the improvement of institutional programs and services. Subjects selected for this dissertation have met the following criteria: (1) Currently enrolled as a 2nd year student at the selected college (2) Will have completed at least 24 units of academic coursework at the selected college by the Fall 2008 semester (3) Are in good academic standing. Recruitment for this dissertation was conducted in two phases, First, faculty members at the selected college who teach intermediate or second year curriculum were asked by the researcher via email to disseminate a letter to their African American students.

The letter introduced the dissertation and stated that if the students fit the described selection criteria and if they were willing to participate that they were invited to contact the researcher in order to set up an informational meeting. This recruitment process was proposed so that the researcher could deliberately gather representation of the African American student population from across multiple academic disciplines.

Secondly, an email invited self-selected participants to an informational session at an off campus location to further explain the dissertation. At that time potential participants

were offered a consent form and a demographic questionnaire designed to gather information about their first year experience. No parental consent was necessary; the student's were all over 18 years of age when the research was conducted.

Instrumentation

This dissertation involved two face-to-face one-hour interviews between the researcher and the subjects. In an effort to provide protection to the subjects, participants selected pseudonyms for themselves during interviews and in the reporting of the data. Moreover, the researcher applied a pseudonym for every instructor, manager or staff member identified in the interview sessions to protect the identity. Lastly the institution will was not directly identified and by holding the interviews in a local off campus location the subjects were further assured an anonymous experience. This researcher utilized this strategy to minimize stress or discomfort to the participants by making the interview environment a conference or meeting room off campus but within close proximity to the college. Additionally the questionnaire was coded for record keeping purposes and remains confidential. The two interview sessions were in the same location but were structured differently. The first interview was designed to obtain information and the second was used to establish clarification of the information obtained in the first session.

Research Questions

- 1. What factors led to the persistence of African American students during their first year at a California community college?
- 2. Have college generated programs or services impacted the success of African American students during the first year at a California community college?

3. What recommendations do current African American students have for the institution and future African American first year California community college students?

The researcher used the questions stated previously in addition to a set of follow up questions in an effort to create a safe and comfortable environment. Depending on the depth of responses given by the interviewees the following follow up questions were utilized. Lastly, additional questions were only asked to encourage further elaboration on a particular response or to provide clarity for the researcher.

Research Question #1: What factors led to the persistence of African American students during their first year at a California community college?

Establishing question: Please describe yourself, where are you from what are your interests and goals.

- a. Please describe how you came to the decision to attend community college, and what where the factors that led to your choice in Cosumnes River College?
- b. What were some of your success and failures during your first year?
- c. Please describe your experiences with college faculty, administrators or staff during your first year.
- d. Please describe your first semester as a student at Cosumnes River College.

Research Question # 2: Have college generated programs or services impacted the success of African American students during the first year at a California community college?

Establishing question: Please describe any opportunities you had to utilize campus services or facilities.

- a. What were your expectations from the college in regards to services provided and your overall student experience?
- b. Please describe your experiences with using technology for school related activities during your first year.
- c. Please describe your perceptions on the availably of campus services specifically for the social or academic support of African American students at Cosumnes River College.
- d. Please describe your social interactions while on campus during your first year at Cosumnes River College.
- e. Please describe your overall academic performance during your first year at Cosumnes River College.

Research Question #3: What recommendations do current African American students have for the institution and future African American first year California community college students?

Establishing question: Please describe what you feel the institution and future freshman could learn from your experiences as a first year community college student.

- a. What helped you successfully complete your full first year at Cosumnes River College?
- b. As an African American student did you at any point during your freshman year feel excluded or not apart of the campus culture?
- c. What recommendations would you give to the institution to further assist African American students in achieving their academic goals?

Data Collection Procedures

Riessman's (1993) research on narrative analysis describes this process of research as unlike other qualitative analysis do to its unique flexibility with no type of standardized procedures for conducting studies. Though the process is not standardized there are steps that guide it as a form of qualitative study. These steps include deciding how to facilitate the interview, the transcription of the interviews and an analytical evaluation of the narratives (Riessman, 1993). This dissertation project collected data through open-ended questioning connecting to the comments given by the subjects.

Narrative analysis is appropriate for this dissertation because of the personal nature of its format and because it can be used to assist subjects in making sense of their experiences through a process of telling their own stories (Bell, 1988).

The researcher crafted three research questions and several establishing questions designed to promote the dialogue and generate an authentic narrative from the participants. The following is an outline of the steps taken to collect the data. Step 1 the college faculty were emailed to request their assistance in the recruiting of participants. Next, the recommended by faculty were asked via email to consider participating, those who self-selected to contribute the dissertation were then contacted. Step 2 included the completion of the pre-interview questionnaire by the students. After the questionnaire was completed students' were then provided a copy of interview questions. Subsequently the students were interviewed and recorded individually at an off campus location.

During step 3 the first round of interviews were reviewed and transcribed by this researcher. Later, the participants were provided the common themes identified from the first round of interviews and a second interview was conducted and recorded. Lastly, the

second round of interviews were then reviewed and transcribed by the researcher. During Step 4 of the process the audio recordings were placed under lock and key and stored at 8401 Center Parkway in Sacramento California.

Data Analysis Procedures

Each interview session was audio recorded and later transcribed by the researcher personally. After the first round interview transcriptions were completed the researcher conducted a critical review of the narratives for relevant themes that were summarized and shared with the participants in the second interview. The interviews from the second meeting were recorded and later the themes from the two narratives were analyzed for commonalities throughout both narratives. The themes discovered in the combined narratives were then used as a basis for recommendations for future research, advice for future African American freshman and for suggested improvements in the institutions ability to serve its African American first year student population.

Protection of Human Subjects

The protection of human subjects is of extreme importance to the researcher.

Therefore the researcher took careful measures to ensure that the participants had complete confidence in the entire research process. The researcher requested and received permission from Protection of Human Subjects committee from the Institutional Review Board for the Protection of Human Subjects (IRBPHS) at the University of the Francisco and remains on file with the University. For a copy of the IRBPHS application, see (Appendix A). Additionally the researcher requested and received permission from the Los Rios Community College District to conduct this research on its student population and remains on file with the district. For a copy of the approval letter,

see (Appendix B).

The researcher provided each participant with a copy of the Subjects Bill of Rights and provided full disclosure of the research intent and process. In an effort to provide protection to the subjects, participants selected pseudonyms for themselves to be identified by in interviews and in the reporting of the data. Moreover a pseudonym for every instructor, manager or staff member indentified in the interview sessions has been used. Lastly the institution was not directly identified in this dissertation and by holding the interviews in a local off campus location the subjects are further assured an anonymous experience.

The researcher attempted to further minimize stress or discomfort for the participants by making the interview environment a conference or meeting room off campus but within close proximity to the college. Additionally the questionnaire was coded for record keeping purposes and remains confidential. The researcher has chosen to conduct two interview sessions. The transcripts and audio tape of the interviews from this dissertation will be kept under lock and key at 8401 Center Parkway, Sacramento, CA 95823 for approximately 1 year after the dissertation has been completed, then the material will be shredded and destroyed.

Portrait of the Researcher

During this dissertation I was a fourth year doctoral student at the University of San Francisco in the International and Multicultural Education department. In addition to being a full time graduate student I was a professor of radio and new media at the institution selected for this dissertation. Lastly, I also served as a part time instructor in

the Communication studies and Bilingual multicultural education departments at a Northern California State University.

I'm an African American male in his early thirties and a former community college student. I'm a Northern California native and a graduate of the California State University system who remains extremely active in the African American college community through his membership in Kappa Alpha Psi fraternity incorporated, a historically black greek letter organization founded in 1911. I received my Masters degree in Education and have used my interest in African American student success and the first year college experience to guide my research agenda. Since graduating with my Masters degree I have given a variety of guest lectures and delivered multiple conference presentations covering best practices in teaching college students, applications for technology in education and the first year experience.

This dissertation was inspired by my experiences as a college student, my close connection to the California community college system and to its African American college student population. As a student I earned a cumulative 1.13 grade point average during my first semester in college and spent my first full year in higher education on academic probation. That humbling experience has shaped my approach to learning and higher education throughout my educational career. I have spent over fifteen years reflecting and studying the elements that contributed to his poor academic performance during my first year and am committed to further exploring the experiences of others as they attempt to navigate intuitions of higher education.

Lastly, as an African American faculty member I was able to create an environment that fostered open and authentic dialogue with the students through mutual

cultural understanding and an unbiased acceptance of the students social and academic experiences. Furthermore, I choose Critical Race Theory as the theoretical framework for this dissertation to provide context for the unique experiences many African American students face when navigating higher education institutions in the United States. Critical Race Theory can also serve as a method of further understanding the social, emotional and psychological state of African American students in higher education. For many African American students, going to college initiates a separation from many of the aspects that build an individuals support system including community and family.

The connection I have with this dissertation and African American students in California community colleges may be interpreted as researcher bias. Yet, in the context of this dissertation I would consider my close connection with the African American student poulation as an added strength. My understanding of the complex experiences of African American students and my presence on the campus provided me the opportunity to establish an environment that allowed for open and honest dialogue about real and valid issues facing this student population. I believe that by utilizing the personal experiences of faculty and students in the planning of services and resources could be used as an example for future researchers and the institution in promoting genuine change within the community college experience of future African American freshman.

CHAPTER IV

FINDINGS OF THE STUDY

Introduction

This dissertation used an analysis of the narrative approach to explore the critical reflections of African American community college sophomores about their first year experiences. This research is intended to provide the institution and future students with information that could be used to improve the first year experiences of African American students as the college continues to grow in population and scale. This section includes feedback in the form of direct quotes from participants in the dissertation in an effort to showcase their individual experiences and provide insight into their feelings about their academic progress and the institution.

This dissertation was conducted through two rounds of interviews that were framed by the questions that guided the analysis. The first round of interviews was used to gather specific information about the participant and their first year experience. The transcripts of this round of interviews were then analyzed for common themes and from that analysis a second set of questions were developed and posed to participants. The second round of interviews provided an opportunity for the subjects to elaborate on key components of their first year experiences and allowed for a chance to further reflect on the factors that led to their persistence. The second round of interviews allowed participants to further dialogue about their experiences as African American students on campus. During this segment of the research process participants included information about their educational history and how past experiences informed how they approached the first year of college both academically and socially.

Profiles of the Participants

The cohort of students used for this dissertation included a total of six participants that consisted of four females and two males. Each student in the cohort completed their freshman year at the institution and are currently full time students that have completed at least 30 units of coursework. As a cohort the students come from a wide array of backgrounds, four of the participants currently have jobs that employ them for twenty hours a week or more and two of the participants hold positions in the campus associated student government. Five of the six participants identify as Christians and one identified that they had converted to an Islamic faith one year prior to this interview. Five of the six participants were born and raised in the state of California, and each of the participants selected the institution based on its proximity to their residence. Four of the six participants stated that they performed well academically during their first semester while the mean of the self-reported grade point averages of all of the participants was 2.79.

The education levels of the parents of students selected for this dissertation was varied and included high school graduates, some college experience, associate degrees, bachelor degrees and graduate degrees. Two of the participants enrolled in and completed the college's freshman seminar course and each of the students surveyed claimed that the college was a culturally diverse and welcoming environment for African American students.

Serena

Serena is a 19 year-old African American female from Los Angeles, CA. She is currently majoring in biology and plans on attending medical school after she transfers from the community college. She currently lives at home with her parents and two

sisters. She commutes to and from campus on a daily basis using regional transit and currently does not have a job. Serena is a first generation college student and currently holds a 3.00 grade point average. She identifies as a politically conservative, catholic and is very involved in her local church. Lastly she is currently an active member in two student clubs on campus.

Eugene

Eugene is a 20 year-old African American male from Sacramento, CA. He comes from an academically weak high school within the local community of the college and worked 15-20 hours per week during his first year. Neither of Eugene's parents graduated from college and he plans to be the first person in his family to graduate from a four-year institution. He is majoring in Business and plans on transferring to a school in the California State University system. Eugene currently lives at home with his parents and commutes to campus daily. Eugene currently holds a 2.30 grade point average and is not actively involved in any on campus organizations.

Iesha

Iesha is a 19 year-old African American female from Sacramento, CA. She has a twin sister and currently lives at home with her parents and sibling. Iesha plans to be the first person in her family to graduate from college and is currently volunteering in local elementary schools in an effort to learn more about the teaching profession. Iesha converted to Islam during her freshman year and became very active in the Muslim Student Alliance on campus and serves as an officer in that organization. Additionally she works on campus as a student development office assistant and remains active as a member of the Associated Student Government. Iesha currently holds a 2.60 grade point

average and is a liberal studies major who hopes to one day become a kindergarten teacher in an urban school.

Samantha

Samantha is a 20-year-old African American student from Atlanta, Georgia who moved to California with her father during her senior year of high school. She is a public relations major, and plans on transferring to a four-year institution to pursue her bachelors of arts and masters degrees. Samantha has earned a cumulative 4.0 grade point average at the community college. She works 25-35 hours a week and holds an elected position in the campus Associated Student Government. Both of Samantha's parents are college graduates and hold professional positions. She lives in her own apartment near the campus and commutes to campus three days a week. Samantha is the oldest of six children and hopes that her siblings follow her example and attend a community college before transferring to a university.

Renee

Renee is a 19-year old African American female student from Oakland,
California. Renee is a business and finance major who plans on working in the banking
industry after graduation. She currently holds a 2.51 grade point average and works 3035 hours per week on campus as an office assistant. During her first year Renee was very
active in student government, the black student union and was a student in the college's
freshman seminar course. She is an only child who still lives with her parents and
commutes to campus five days a week from her home. Both of her parents hold college
degrees and she comes from an academically high performing high school.

Q is a 19-year old African American male student majoring in television production. Q was born in Los Angeles, California and relocated with his family to the Sacramento area when he was in elementary school. He comes from an academically strong high school and graduated with a 2.35 cumulative grade point average. In high school Q reflects that he often did the bare minimum academically and would often only be motivated to perform to stay active as a member of the golf team and as a member of the high school choir. He holds a cumulative grade point average of 2.67 at the college and is a member of the college's golf team. Q aspires to transfer and play golf at an NCAA division II institution and eventually become a sports broadcaster. Q works 15-25 hours a week, commutes to campus three days a week and lives at home with his parents. Table 4 provides a demographic synopsis of the participants selected for this dissertation.

Table 4

Demographic Data of the Student Participants

Student Name	Age(s)	Gender(s)	Declared Major	Grade Point Average
Eugene	20	Male	Business	2.30
Iesha	19	Female	Liberal studies	2.60
Q	19	Male	Television production	2.67
Renee	19	Female	Business finance	2.51
Samantha	20	Female	Public relations	4.00
Serena	19	Female	Biology	3.00

Findings

Research Question #1: What Factors Led to the Persistence of African American

Students During Their First Year at a California Community College?

This dissertation provided an opportunity for subjects to reflect and dialogue about some of the factors that lead to their academic persistence during the first year of college. Throughout the conversations about their freshman year of college the first research question generated two very distinct subthemes: (a) Making personal connections with college faculty effects the overall learning experience; and (b) Involvement in campus activities can lead to a stronger connection with the campus and serve as motivation for matriculation.

Personal Connections With College Faculty

All six of the students stated that making personal connections with professors' added value to their overall educational experience during their first year. Though each of them had unique experiences with professors across multiple disciplines the common theme that ran through all of their reflections was that when a professor showed a personal interest in their education it motivated them to care more about their own progress. Through the interviews conducted in this dissertation students' reflected on beginning college without having a connection to the faculty and how this significant shift from the familiarity of a high school setting lead to feelings of isolation and anxiety when they first arrived. The following quote was generated from Samantha when asked about her interactions with professors once she arrived on campus and started classes.

Samantha explains:

They were all very, very helpful. One in particular I ended up spending a lot of time with; he's become somewhat of a mentor, very, very

helpful. Another one of my professors who is also my advisor with Student Government was wonderful. I've had like a really great experience working with my professors. I just love the professors and the environment and how you get a more personal one-on-one connection with your professors than at a four-year university. They make me want to be here and do well.

Q also comments on how making connections with instructors helped him during his first year. "The community college level and the faculty at the college have really, really helped me to become the person I am today in taking my education more seriously." These statements from student's support the idea that making a personal connection with the instructor can provide much more than motivation to attend class, a connection can provide students with an emotional tie to the institution that further links them to the learning process. A connection with a faculty member can also serve as an opportunity for student to learn first hand about the differences between high school and college. Iesha explains how as a student a faculty member may be able to help you change your thinking about being successful in college:

To be a student, you have to have the right mindset. You've got to be here to actually want to do something. A professor told me once that if you're just here to waste your time, you can do that anywhere else and if you're going to do that you might as well just do it somewhere like a some dead end job where you're making money rather than just wasting your time and hanging out and doing nothing. You need to be able to conquer other stuff that's going to come your way and really focus on the education side and embrace what you're learning. I'm sure that there's probably some really good information that I should have learned and I should have understood from my first two semesters and high school as well, that I just didn't.

Renee's perspective illustrates how the influence of a professor can have an immediate effect on students, she explains:

My first year, my professors were great. I had one particular teacher, she was my freshman seminar teacher, she was awesome. Like we couldn't really understand her but we got her points by the way she described things, it made you want to go to college. It was a freshman seminar class and she wanted you to succeed here. But she really taught me a lot of things, especially about time management. I was

always so busy so I never took the time to say, this time is for me and I need this time for my books and this time for study and this time for work. So if it wasn't for her, I probably would still be struggling.

Serena had varied experiences with professors, some interactions were extremely positive and some created new challenges. She states, "I did not choose maybe the right teacher and it was hard for me to understand his class. Then when I went to the Math lab it really did not help as far as it was too two different teaching styles. So the lesson I learned was to make sure I choose the right Math teacher." This experience has lead to Serena's decision to learn more about professors before she enrolls in classes. The practice of selecting professors based on reputation and past encounters is very important in creating the best possible scenario for academic success. In addition to using personal experience to determine what professors to select as instructors, students also utilize peer networks and in some cases the Internet to evaluate their probable success based on the reputation of the professor.

It's clear that after analysis of the transcripts from both rounds of interviews that contact with professors both inside and outside of classroom is just as important as contact with other students. Professors can serve as an integral part of a student's education through work as mentors, advocates and as resources in navigating the matriculation process. Furthermore social interactions with other students can serve as a means of peer mentoring and peer support systems.

Renee reflects on how she felt about college during her first year and what type of connections with school she was accustomed to:

My first year here was pretty difficult. I really didn't have my mindset to attend school, but in order to stay on my parents' benefits I had to go to school. So I think for the most part, coming to college and really trying to do the transition from high school was really hard for me. I was making great money working for

the bank and I enjoyed it. I worked there during high school, too, so I enjoyed it and that was the only thing I really knew or really that got me by in high school was having a job and making sure they checked my grades and all that things. When I got to college, my first semester, no one did those things and I think I kind of fell through the cracks, like oh, I don't have a bell to get me to class or I don't have anyone to remind me to go to school. So I was content just working and taking one or two classes. It was so convenient, no one told me, you have to take this and you have to take that, so I just chose the easiest classes just so I could work more hours to make money.

Eugene reflects on the transition to college from high school, "First year, like I said, I was mainly trying to get a feel for the college experience but at the same time I had a lot of other things going on as well." These subjects' provided insight into one of the issues that may face future African American students when making the transition from high school to college. Three of the six participants spoke of having the right mentality for college and how for them being mentally prepared was the most challenging aspect to performing well as a student.

Serena explains:

I'd tell the incoming freshmen to you know, not only ask for help if you need it but anytime to get to know the professor and especially when you need to find out something. But also come onto campus with a certain type level of courtesy and intelligence because now that you're coming from high school, you are legally an adult and you should act like one. Because when you come to the campus you have to hold the mentality of being a college student because that's what's expected of you. If you don't then you will not survive here long or go on to mentality of a college student, otherwise you'll stumble and fall.

The mentality of the student as referenced in this quote highlights many of the challenges facing incoming students. Not clearly understanding the difference between college and high school can leave some students confused and lacking in the necessary skill sets to connect constructively with faculty. The lack of these college social skills can cause setbacks to a student's overall acculturation to college life and their progression academically by isolating them from the networking opportunities engagement with

faculty can provide. The subject Eugene explains his personal feelings about approaching college from a mentality that is open to social interactions and showing the initiative to make connections with his professors. Eugene explains:

I was an interactive person in high school, so this to me was nothing new. I'm not really shy, so talking to people and socializing with even professors isn't hard for me. I think that when you socialize a lot more, you find a lot of different network opportunities and then it makes everything come together when you're in college. Because if you need help with something and you talk to this person that can help you with it, you can just go straight to him and get it. Instead of you know, if you're quiet and shy, don't want to talk to anybody, it's kind of harder.

The right mentality and making personal connections with faculty comes up again with Iesha. She explains:

Okay. So my first semester, I came with the high school mentality. Because while I was in high school, I had a lot of freedom; I was involved in every club and I really didn't put much effort into my schoolwork. So I felt like I would be able to come over here and like so-called run the school as I did in high school. Like just tell the teacher anything and they'll believe me and I could just get by with doing halfway done work and all this stuff. So I was kind of cocky and arrogant, I guess you could say and I'd always be running around with my friends. I remember leaving campus all the time, or, I don't need to go to this class, or, that class is not that important and stuff. So I just kind of had that mentality of just like I'm here, whatever, school's not that important and I'm here to socialize type thing. And so yeah, I hit a wall really fast, and I figured out that you have to come with more respect for what college is and how your professors can be cool.

These students have been affected in positive ways by constructive social interactions with their professors and these contacts have made a significant impact on how they approach college. The relationships established between the students and the faculty has given way to new opportunities for the students to benefit from the knowledge of the instructors beyond the subject matter they teach.

Five of the six African American students in this dissertation had different relationships with their African American professor that with their non-African American professors. Throughout the conversations the participants expressed a distinct connection

with faculty that shared cultural norms, those who more effectively communicate with African American students and those who might have struggled with similar issues during their time as students. Iesha explains her feelings towards African American professors:

I wouldn't say that my interactions are extremely different one minor difference would be that we speak to each other. You know like they understand what I am saying and I don't always have to speak proper. There's that internal understanding, we understand each other because of our ethnicity and some of the experiences that go along with being black. We are a lot more comfortable with each other. With my non African American professors there are some boundaries that I would not cross with my African American Professors. Because it's more acceptable and its just more natural the openness and the warmth that black people share with one another it just like that.

Renee reflects on her feelings about the difference in working with an African American professor compared to working with a professor from another ethnic background. She explains:

I think that if you have a teacher that is of the same race, I think you feel more comfortable with them in knowing that they are out to help you. And they have some of the same experiences, whether it was racial differences or any cultural... anything. I think it works out better because I mean, a Caucasian teacher can't really say, "Well, I had that same experience," because they are white.

Eugene also shared his experiences with African American faculty and theorizes why their presence on campus is so beneficial to African American students. He explains:

I would say we connect more only for the simple fact because there's not too many on campus and there's a lot of African Americans here. So when we did see some, or see African American professors, the relationship between us is more... I wouldn't say friendly, but it's more understanding, in a sense.

From the evaluation of these transcripts it's clear that the interactions these student's have with professors that share cultural heritage or shared experiences supports the theory that students can benefit from a diverse faculty and the personal experiences of professors who may have faced similar challenges through their academic journey.

Involvement in Campus Activities Can Lead to a Stronger Connection With the Campus and Serve as Motivation For Matriculation.

As the participants discussed the first year experience, the concept of campus involvement became very prevalent. Though all of the participants mentioned involvement in their reflections about the first year. Three of the six reflected on how being involved on campus with organizations or services played a major role in their progression through the first year. Their experiences with being involved on campus were framed either by an initial lack of involvement during the first year and how they later recognized the value in participation, how being active on campus can serve as an opportunity to gain access to new resources, or the reflections reiterated the benefits associated with being active in some sort of campus organization or club. Samantha explains how not being involved initially changed once she recognized the value in student organizations.

I think that not just for African Americans, but I think students in general, though, need to be more involved with each other. I think that unity, coming together and helping each other even if it's just African Americans or any group at all, people need to come together. I think that the campus itself, whether it be like a certain club or a certain group needs to find a way to bring all the students together and show that... along with the professors and that they have support and that if they need anything, that our professors are there. Like we have a Welcome Day, but usually on a Welcome Day we just give out free food or a snack or something. But I think maybe there needs to be something where we have activities, like a get to know each other activity and get people's energy up and their spirits up. . Because like I said, I think that involvement and having a professor that can help you or a mentor in your campus with you really, really helps, because I honestly don't think I would have been able to do it without the involvement on campus or without my professor.

Samantha's statement speaks to the connection some of the subjects had with the institution based on an involvement and how it can generate opportunities in bringing students together from varied backgrounds and experiences. Through the reflections of

the subjects the type of involvement a student can participate in can be very diverse. Student involvement could be interpreted in a number of ways including participating a club, being a part of college athletics, joining student government, utilizing extended opportunity and programs and services or developing a mentor/mentee relationship with a professor or coach. During the interviews Serena reflected on how becoming involved changed her college experience, she explains:

My first semester I did feel kind of excluded because I was not a part of a whole lot of everything. I didn't join MESA (Mathematics, Engineering, Science Achievement) until the end of the semester, my first semester. The second semester, that's when I started doing a whole lot of things. I got involved in the Science Club and MESA more where I was going near every day now, five times, five days out of the week. And being around more that, I got more intertwined more with the society on campus. So the first semester I was excluded because I didn't get into anything and I was just by myself. The second semester kind of opened up a whole new door when I started getting involved in more things.

Iesha was able to overcome some serious challenges during her first semester in college by becoming involved in some of the student service opportunities provided to students by the college. Iesha explains:

I started to learn because teachers would put emphasis, like, "Aren't you enrolled in the Reading and Writing Lab?" I'm like, what is that? So I would have to go take a look and see what it was and like it's a great facility for students to use. The thing I think most students, we don't get involved in a lot of facilities on campus or the resources because we don't know about them and the reason we don't know about them is because we're not paying attention. Getting involved in our services on campus is so important to being successful. It really helped a lot.

All of the participants discussed involvement but there was a difference between how the female students seemed to value involvement and how the male students valued being active or connected to the campus through extra curricular actives. Additionally neither of the male participants focused on attending college directly after high school.

Based on the dialogues with these two individuals it I would argue that the lack of

involvement on campus was increased by their hesitancy in attending. Eugene had planned on joining the United States Marine Corp before he ended up at community college and from his own admission this contributed to his less than enthusiastic interest in attending school. Eugene states, "First off, I was not prepared for college. Through high school, my main focus was not college. I didn't really have no perception of college because like I said, that wasn't my focus right out of high school."

Q was also initially hesitant in attending college after graduation, he explains, "Out of high school I really didn't have the head for college, like I said before, I never really have been a good student." Based on information gathered through the pre interview questionnaire Q's only high school commitments were the black student union and the gold team. Therefore it is likely that the opportunities available to him as a first time college student could have been ignored or he may have been oblivious to their existence.

Q's limited involvement with the campus outside of his academic requirements was confined to playing for the golf team. His interactions with classmates were limited to in class activities and he reflects that he would often spend little time on campus socializing. Eugene has approached becoming more involved in the college experience from a social perspective. He makes reference to spending time on campus outside of class as a social exercise with no established structure or purpose. Eugene's campus involvement was limited to hanging out with friends and though this is a valid reason for spending more time on campus he was unable to associate that type of interaction as beneficial to his first year success. Though Eugene was unable to make the connection between hanging out on campus and persistence, social involvement into college is a key

element in making students feel as if they are a part of the institution. This integration or involvement in the campus macro culture can occur through communication with faculty or campus services (Tinto, 1993).

Research Question # 2: Have College Generated Programs or Services Impacted the

Success of African American Students During the First Year at a California

Community College?

The participants of this dissertation reflected on their experiences during the first year at a CCC and provided insight into what campus programs or services assisted them during their freshman year. The central theme that emerged from the dialogues regarding college programs and services was participation promotes a feeling of support and generates a stronger connection between the students and the institution. Three of the six participants reflected about active participation in a variety of campus sponsored programs or services and validated their interaction as critical in their progress through the first year.

Iesha reflects on her participation in and discovery of campus services:

The opportunities that I've had, they were presented to me but at first I really didn't embrace them, I didn't understand them, I didn't absorb what they were. So my first semester I paid for everything and that was probably like... Yeah, I paid for it, for sure. Then I started to learn what was EOP was. I had no idea that the college could give you money. It's probably not because it wasn't presented to me, it's just because I didn't absorb it—I didn't understand what it was. But then I started to learn because teachers would put emphasis, like, "Aren't you enrolled in the Reading and Writing Lab?" I'm like, what is that? So I would have to go take a look and see what it was and like it's a great facility for students to use. The thing I think most students, we don't get involved in a lot of facilities on campus or the resources because we don't know about them and the reason we don't know about them is because we're not paying attention. Getting involved in our services on campus is so important to being successful. It really helped a lot.

Iesha was not the only student who later discovered opportunities for student support provided by the campus. Samantha reflects on her experience:

Everything you need is here—like computer labs, for example. My favorite computer lab is the one in the library because it comes with headphones so while you're working you can listen to music on Youtube or MySpace or Rate My Professor.com or whatever. There's also... That lab that closes at 8:45 or 9 o'clock and the other two are in the specific... in the LRC. But there's plenty of computers for everybody. There's also a Math tutoring lab that you could enroll and it's like half a credit. You can go in there and get help from professors at any time. There's also the writing lab where they help you with your writing. There is the lending library, there's Financial Aid, which if you have any questions, you can ask them there. There's the cafeteria, there's... I mean, I used like most of them my first year and some of them I wasn't even aware of until this semester.

The variety of services combined with the access students have to technology creates an environment on campus that empowers students with support in many of the basic skills requirements for college. Computer labs are incredibly important to the academic success of first time students at the community college level. Beyond the social networking applications and the wealth of knowledge provided by the Internet, computer labs on campus allow socioeconomically disadvantaged students to have access to the tool necessary to complete their educational goals.

Serena's perspective on the availability of campus services and how they would impact her academic progress was influenced by her high school experience. She explains:

I expected them (the community college) to have some sort of help there as much as possible. I'm using the model from my high school... I mean, we had a computer lab and we had the library as well. We had services on campus to be able to help us to some extent. I was always in the library in high school and I had some help maybe from the librarians but I often gone to see my teachers after school and asked them as well to set up things to see if I could help... get some help with that. So in ways, yes, I expected at least that. Especially the technology stuff.

Serena's experience highlights possible expectations of students from academically strong or technology proficient high schools. She also reflects on participating in the reading and writing lab on campus, "Some of my successes were being able to adapt to the teachers' teaching styles and to be able to have access to the library and reading and writing center and especially my fellow students as well." Serena's participation in student services not only provided her with a means to understanding complex issues; it also provided an opportunity to work with other college students in a supportive and structured academic environment.

Those participants who elected not to participate in campus services during the first year still recognized their value and at times served as advocates for them within the student population. As a student assistant Renee reflects on her understanding of the services and systems provided by the institution and how they can work to support students as they navigate the community college system. "I haven't been one to use any of the facilities on campus, but I am an advocate for them; I tell students where they're offered." Renee's reflection is an example of how student recognize the benefits of the programs and services being offered even if they do not choose to participate in them. Research Question #3: What Recommendations Do Current African American Students Have For the Institution and Future African American First Year California Community College Students?

Three distinct themes were identified in the review of the transcripts based on this research question. The student's recommendations for the institution as they relate to serving the African American student population could be isolated as one central theme.

This theme included; (a) encouraging instructors of all backgrounds to build relationships

with the African American students they encounter. The refection's of the participants echoed the sentiment that African American students are a unique population and as such may need to be encouraged more often in order to succeed at the rates of other groups. Additionally the reflections produced two themes as recommendations for future African American students; (a) social integration with the campus culture; (b) developing a personal relationship with your professors. All six of the participants reflected on the effects of becoming integrated into the culture of the campus and how that experience coupled with personal relationships established with professors provided additional incentive and insight into persisting academically and growing socially while attending the community college.

Institutional Recommendations

Each of the students used as subjects in this dissertation had suggestions for the institution in its growing efforts to serve the African American student population. Renee gave a strong message regarding the perceptions of the institution towards African America students. She explains:

I believe that professors and faculty, administrators and staff, need to understand that students, African American... And I mean not just African Americans but particularly them, they have struggles at home, they have struggles in other areas. So they come to school on a daily and they need to talk to someone or you see them struggling, when you see this, take the time, talk to them and actually see what's going on with them.

Serena's comments directed towards the institution focused on when African American students are recognized on campus and the effects outreach can have on a future student. She explains:

Well, I have to say two things to them as far as that. As far as for African American students or black students, if you will, I would say not just do it just once a year as far as only one month when it's just 28 days, do that. Why not do

it at least maybe once a month or once a week, if possible, to show that African American students or black students, you could say, have a great chance of accomplishing or doing anything for that. And that kind of inspiration can instill into them... Or like another thing I would suggest for them is to go out to some of the schools and have them do presentations as far as having one of the CRC representatives come to the school and show them. Actually have a black representative go there and tell them that the can succeed at a community college.

Serena's message is simple and logical; if the community college truly wants to reach out to African American students then there must be more than black history month programs and general outreach attempts. The exploration of the contributions of African American must be ongoing and the outreach of African American students in high school must be targeted and multifaceted.

Samantha recommendations for the campus are directed at the college's ability to establish and support opportunities for student engagement. She explains:

I think that the campus itself, whether it be like a certain club or a certain group needs to find a way to bring all the students together and show that... along with the professors and that they have support and that if they need anything, that our professors are there. Like we have a Welcome Day, but usually on a Welcome Day we just give out free food or a snack or something. But I think maybe there needs to be something where we have activities, like a get to know each other activity and get people's energy up and their spirits up and notice hey, you know, wow, there are a lot of professors here and this professor... Like even have a couple of professors speak, "Hey, I'm so-and-so and I teach this," etc., etc. Because like I said, I think that involvement and having a professor that can help you or a mentor in your campus with you really, really helps, because I honestly don't think I would have been able to do it without the involvement or without my professor.

Samantha in this statement makes an excellent case for faculty participation to extend outside of the classroom and for the college to become an instrument for not only intellectual growth but for social growth as well. Eugene's comments not only apply to faculty of the institution, he recommendation speaks to a possible need for the hiring of more African American professors campus wide. He explains:

Basically more African American teachers should be more active with African American students. Hands down, period (laughs). Because like me for example, I met a teacher and as a black man he has really given me a new person to look up to. I mean I didn't know that black professor were up here until him, I mean I barley had ever seen black male teachers and couldn't really imagine one at college. And he's down for us, black students and he's cool, not just some black dude acting white. He's really active with me, I talk to him, e-mail him and that helps me a lot because it gives me a different outlet as far as teachers. Not just white American teachers but black American teachers. And beside you feel more comfortable with black teachers, you kind of share a certain vibe with them so it just makes it better. So African American professors should be more active with the black students.

Q is an example of a student who has also benefited from the attention of an African American professor. He explains:

I'm not saying that maybe whites don't have a hard time but like bringing personal views can help black students relate to college. A black professor of mine gave us actually a big insight, which really opened my eyes to where I am now. He talked about, I believe it was the first or second day of class, if you remember, about his majors. About the GPAs the first semester and everything like that. And then that really opened my eyes like wow, he's really pretty much must like me. Dead on me, basically. So if he can do it, I can do it too. I used to really doubt myself a lot especially in the past. I've been getting a little bit better, my professors have been helping me about that throughout the year.

Iesha's reflection about what the institution can do to further assist African American students is clearly reflecting the idea that college cannot be easily navigated without direct support from the institution. College students in general and African American students specifically face varied challenges and obstacles in achieving their academic goals. She explains:

I don't know how to explain it, but I just think that just to treat them like students and not to belittle them in any way and to embrace the things that they are good at and to really push towards those things. Because I know for a fact that I feel like as a black student, we work harder in class and we like to engage in conversations and we like to participate, but how are we going to get motivated more at home?

Social integration with the campus culture

The participants reflected on getting involved on campus and how being connected to the campus in some way in addition to takings classes provided them with more motivation and resolve as first year students. Students are not relegated to playing sports or joining clubs, there are student services and programs that have been established to aid in building a tangible connection to campus, that can be so often missed at commuter campuses. Renee reflects on one of the advantages of being more involved on campus and how those actions can provide new opportunities. She explains:

Right now I would say being involved is most important, I think that, it's proven that if you are in a college setting and you're more involved, you're going to do better in school. So if you're not involved you tend to be more... not relaxed, but you tend to be more shy and not willing to get help or go into the office hours. But if you're more involved you understand the opportunities that are given to you so you're going to take more advantage of them. So it works out.

Promoting campus involvement can also come in a variety of ways, but after reviewing the transcripts it becomes clear that it's important to establish for first time freshman clear limits in order to prevent a situation where academic progress is affected by the student's commitment to campus activities. There is no perfect balance or equation to this effort, yet it should be noted that the social connection that African American students make with an institution could be related to their overall persistence.

Developing a Personal Relationship With Your Professors

Iesha reflects on how important it is for freshman students to connect with their professors outside of the classroom encounters. She explains:

Yeah, it's really important to know your professors, let them know you because they see hundreds of faces and you may see them and you think that oh, yeah, they know you but they won't know you unless... You have to make sure that they know your name so that when they see you in the hallways they do say something to you. You want to get that kind of connection with them to where

when they're grading your paper they're thinking about you not just like man, who is this? Just trying to figure out who this is and then it's just a paper. But when they're grading a paper and they understand who you are, the way you speak, the mannerisms that you have, it kind of makes it more real to them and they may or may not be able to understand better what your purpose is or what your goal is, what you're trying to get across to them through papers or assignments or whatever it may be.

This statement and previous reflections by the participants highlight the critical need for freshman students to recognize one of the fundamental differences between high school and college. Within higher education there is a greater chance that instructors are less likely and in some cases not willing to exert themselves in getting to know the individual needs and goals of students they encounter. Therefore if an African American freshman that seeks to establish a connection, may in fact need to initiate the contact and seek out mentorship. Eugene expresses his recommendation for incoming African American freshman. He explains:

Coming from high school, it's kind of like the teachers have to come to you, in a way. Its there job to know you and plus they see you all the time. So my advice for incoming freshmen, first year of college, just be ready to go *to* your professors. Don't just wait for them to come to you. Don't be afraid to ask questions, and trust me get help with the work its way different.

Eugene's comments are not rare among students who have successful navigated the first year of college and who have found support in professors and mentors. Making connections with professor is not an absolute solution to the academic and persistence issues facing African American students at he community college level, yet perhaps this can be a starting point for realizing real change within one of the most valuable education resources ever established.

Summary

The critical reflections of African American sophomores about their academic success during the first year, has yielded a number of opinion of what is means to succeed academically. For many of these students simply remaining enrolled in college while many of their friends and family members have halted or discontinued their pursuit of higher education is an accomplishment in it self. Additionally through the sharing of both successes and challenges these students have provided insight into a population that has historically been challenged in reaching proportionate rates of college completion compared to both majority and minority student populations.

The participants all reported that the campus climate and campus culture made them feel comfortable and that at no time did they ever feel excluded. The college generated a comforting environment or opportunity for equal access to learning resources. One student reflects on how included she has felt. "The campus itself is very diverse and I can't… No, I've never felt excluded anywhere, I always feel very welcome." Each of the participants in this dissertation shared a unique and important first year experiences. These reflections are based on a variety of factors and though there were common themes there were individual experiences that added diversity and depth to this process.

The following chapter includes a summary of the dissertation and conclusions based on recommendations for future research and the institution. This dissertation also generates new insight into a sub-population of current research already conducted on the first year experience and broadens that body of work to include community colleges as well as four-year institutions.

CHAPTER V

DISCUSSION, CONCLUSIONS AND RECOMMENDATIONS

Introduction

This dissertation utilized analysis of the narrative to explore the critical reflections of African American community college students about their successful transition from the first-year to the second-year of college. The analysis of participant's reflections allowed for a rich and personal examination of the complex issues around first year African American student academic success. This dissertation serves as a resource for the institution and for the district in its' efforts to reach the African American student population. By utilizing the students in the examination of persistence and retention issues, this dissertation stands to provide new perspective and insight to current practices. For the purpose of this dissertation Critical Race Theory was applied as a part of this dissertation in reference to the historically unequal circumstances African American students have faced over the history of the American educational system.

Currently within the state of California there is greater access to higher education than at any other time in history, yet at California community colleges African American students achieve at substantially lower rates that all other students populations (California Community Colleges Chancellor's Office, 2007). This disparity in success rates between African American students and other student groups provides an opportunity to approach the problem of student success a valid topic for scholarly exploration. The research conducted through this dissertation candidly includes and analysis the experiences of African American students that have successfully matriculated from the first-year to the

second at a Northern California community college and identifies their recommendations for institutional change and for future student populations.

This dissertation investigated a cohort of six African American students who are currently in good academic standing and have attempted and completed their first year at the community college level. The research sought to explore the following questions and identity potential strategies for supporting and engaging African American community college freshman:

- 1. What factors led to the persistence of African American students during their first year at a California community college?
- 2. Have college generated programs or services impacted the success of African American students during the first year at a California community college?
- 3. What recommendations do current African American students have for the institution and future African American first year California community college students?

Two one-on-one interviews were used to gather perspective and insight into the experiences of African American community college freshman. These interviews provided an opportunity for students to dialogue about their personal stories of success and failure in a context designed to identify applicable strategies for success. The transcripts of these conversations were then analyzed to and used to unearth practical strategies that could be utilized to promote improvements in current persistence and retention rates. Furthermore, the student reflections revealed strategies for success that allowed them to complete the first year of college and provided warnings for potential

pitfalls facing future students as they enter higher education through the ranks of community college.

Discussion

Based on the design of the dissertation, the discussion is organized around exploring the three research questions that guided this dissertation. The responses of the participants were explored in an effort to identify noteworthy themes that may have emerged from the dialogues. The themes that have been identified represent the perspectives of African American community college students who have all persisted to the second year and continue towards their educational goals in good academic standing. Research Question #1: What factors led to the persistence of African American students during their first year at a California community college?

Two principal themes emerged from this research question (a) Making personal connections with college faculty and staff; and (b) Active participation in campus clubs or organizations are critical elements to the continued persistence of African American students. During the reflections the participants were open about how important becoming a part of the campus community was to how they perceived their academic experiences. Additionally each of the participants remarked on how essential it is for African American students to take advantage of the opportunities for engagement provided to them by the institution and to seek out support from faculty and staff at the very beginning of their college career.

Personal Connections

All six of the participants made comments in their reflections, that establishing a personal connection with a professor of staff member had a significant impact on their

desire and ability to persist through the first year of college. Participants reported that the benefits of making personal connections with faculty or staff included extra help with course work, first hand mentoring about the higher education process and the establishment of special connections to the institution.

Five of the six participants made remarks to how their interactions with African American professors compared to non-African American professor generated a deeper sense of support and guidance as first time freshman. Crosson (1988) argues that the racially hostile environment colleges can breed make it vital for African American students to have contact with college faculty that may be more attuned to their The primary difference between interactions with African American professors and non-African American professors as presented by the participants was in how African American students and faculty related to one another through shared experiences and understandings. Having African American professors on campus that African American students can build relationships with is a critical element in providing support to students as they navigate the communities of academia and home (Jalomo, 1995).

By interpreting these reflections using Critical Race Theory as the established framework race is still a relevant issue in the overall academic and social success of African American students. The participants stated that African American professors had successfully navigated college and were willing to share their own experiences and recommendations with students in an effort to help the better understand college as an institution. According to the students' African American professors were able to provide mentoring from the perspective of a shared cultural background with the subjects and this was seen as both helpful and inspiring by those who participated.

The first sub-theme that emerged from these reflections identified making social connections with other students was critical in providing an environment of peer mentoring and support. This social connection also serves as a form of further motivation for students to become more aware and accountable for issues relating attendance and academic progress. Additionally this form of interaction served students by helping them develop social skills and a social network that proved useful when selecting courses and spending extended amounts of time on campus. By interacting socially with other students, participants reflected that they felt more apart of the campus culture and consequently instilled in them a sense of school sprit and pride in being a college student.

The second and final subtheme that emerged from this area indentified that students' must discipline themselves mentally for the rigors of higher education. Three of the six participants reflected about not initially being mentally prepared for college and how their lack of understanding about how different college was from high school had an impact on their academic progress during the first year. They commented that high school was a place that they were comfortable and they knew what was expected of them because they had been around that style of schooling their entire lives. Through reflections the participants commented about having all of their educational decisions made for them in the k-12 system. Their books were loaned to them, their schedules were predetermined, their free time was scheduled and their access to the outside world was limited by closed campuses and bells that told them when to attend and leave class.

When they attended college for the first time, participants reflected that they weren't sure on what classes to take, they were not clear about how to get textbooks and

they were surprised that they were ultimately responsible for attending class and how much time they spent on campus. Participants eventually used social connections to acclimate themselves to differences between high school and college. Furthermore these connections with other students or with faculty became key in assisting the participants with making it through a very difficult transitional period.

Personal connections with others associated with the campus were cited as being important to all of the participants. Samantha reported that her connections with a professor made her want to come to school when she was struggling in a course, that connection has since developed into a mentoring relationship. Renee has benefited from her exposure and relationship with college staff provided through her involvement in student government by finding on campus employment that is flexible around her school schedule. Eugene has used his personal connections with faculty and peer networking in evaluating what instructors to take and what path he should follow in regards to transferring to a four-year institution. For Q, the relationships he has developed with professors have afforded him important insight into local and regional universities he was considering as destinations after community college.

Involvement

Three of the six participants self identified themselves as being highly active in campus events or programs. These students were involved in a variety of programs including student government, clubs and college athletics. They credited their involvement in these extra curricular activities to providing additional motivation to attend classes and achieve academically. The three participants that were not actively involved in the campus stated that they recognized the benefits of involvement and if they

were not associated with clubs or organizations on campus during their first year they made an effort to get involved as sophomores.

A form of support and influence came from students who were involved with student government as freshman. More than the typical student experience these participants were consistently exposed to peers who were student leaders on campus. Additionally these participants' had exclusive opportunities to work with college administration and faculty leaders as part of the colleges shared governance structure. As a member of the student government program students were also exposed to course in leadership, and peer conflict management. These participants also were able to attend conferences and workshops that allowed them unique opportunities for personal and professional development. Several of these participants commented on how this access affected their priorities and their academic goals as they pertained to transfer and their desire to attempt to earn advanced degrees.

Participants were very candid about their experiences with being associated with campus organizations including the Black Student Union. Though each of the participants recognized that the organization existed none of the participants were currently involved. The participant's reflected that they felt that the Black Student Union could be an organization on campus that immediately connected with African American first year students and should be presented as an opportunity to develop connections with faculty and staff of color. Unfortunately none of the participants' felt that that the club was reaching it's full potential, therefore it was losing out on the energy and ideas of new members. In some reflections the participant's negative perceptions of the Black Student Union were directly associated with the lack involvement of African American faculty

and staff. Others believed that it was the responsibility of the African American student body to organize and empower the club. One participant reported that he was active in the high school version of the Black Student Union and based on the lack of visibility of the club in college he felt that it couldn't have been the same type of organization. Each of the participants remarked that they felt that the significance of the organization lied within its social networking potential and its power to connect with African American employees of the campus with African American students.

The research has shown that students who participate in organizations and programs are more successful and make more meaningful connections with the institution compared to non-participants (Astin, 1984, 1993; Berger & Milem, 1999; Crosson, 1988; Ford, Obiakor, & Patton, 1995; Tinto, 1993). In addition to the Black Student Union the participants discussed, student government, athletics, student sponsored clubs, reading and writing centers, the extended opportunity programs and services office and the mathematics engineering science achievement program as opportunities in campus involvement. Astin (1984) suggested that when college students are involved in campus life and cocurricular activities that they are more successful. The findings of this dissertation support that assertion. Being involved in some part of the campus in addition to academics has led to the personal development or academic achievement of all of the participants in this dissertation. Participants reported that being involved in the campus has added value to their overall educational experience and has made an impact in what they hope to achieve in subsequent life after community college.

The participants of the dissertation were complementary of the college faculty, staff and administrators they had the opportunity to interact with during their first year.

The college campus was described as a welcoming and supportive environment that provided numerous opportunities for students to engage in activities and be intellectually stimulated. Each of the participants commented that they had never felt excluded as minorities or discriminated against because of their race or gender. They all reflected on how the college worked to acknowledge a variety of cultural groups through workshops, fairs, guest speakers and special programs therefore living up to its well publicized commitment to cultural competence. Samantha reflected that the campus was a place where she felt safe to be who she was and was a place where she felt she could grow as a person both academically and socially.

Research Question #2: Have college generated programs or services impacted the success of African American students during the first year at a California community college?

College programs and services are designed to provide support and assistance to students during their college experience. Programs and services offered by the community colleges can include financial aid, disabled student services, veteran services, academic and psychiatric counseling, computer labs and tutoring opportunities. These programs are identified as student services and serve in combination with instructional services a comprehensive learning environment.

Three of the six students reflected that they were actively involved with student services during their first year and that these services provided them with the support they needed to finish out the year. Iesha commented about her experience with student services during the first year as an opportunity to expand her understanding of how colleges instituted opportunities to assist students. Samantha also dialogued about the

discovery of student services during her first year. In her case the availability of on campus computer labs with high speed internet provided her the access she needed to complete homework assignments and network socially via the world wide web. Serena utilized the campuses reading and writing labs as an opportunity to gain help with her English course. She reflects that the lab allowed her the extra time she needed to better understand what her instructor was lecturing about in class. Additionally the labs provided and opportunity to associate with other students and form peer tutoring and support mechanisms.

The students' who did not take advantage of student services during their first year all reflected on their availability and eventual involvement in student support programs. Additionally participants' reflected on the importance of spreading the word to other students about to opportunities for student support offered by the campus and the benefits of taking the time to get involved early in the academic process.

The central theme that emerged from the dialogues regarding college programs and services was participation in student services promotes a stronger connection to the institution and provides support during a period of transition. Three of the six participants reflected about active participation in a variety of campus sponsored programs or services and validated their interaction as critical in their progress through the first year. The numerous services offered by the campus combined with the access to technology, establishes a supportive network on campus that allows students access to basic requirements for college success.

All six of the students commented on the importance of having access to computers during their first year. The computer labs on campus are incredibly important

to the academic success of first time students at the community college level because many of the instructors have web-based content and the college utilizes the internet as a means of registration and communication. Beyond the social networking applications and the wealth of knowledge provided by the internet, makes computer labs on campus a link to technology for socioeconomically disadvantaged students who must have access to the tools necessary to complete their educational goals.

Research Question #3: What recommendations do current African American students have for the institution and future African American first year California community college students?

This research question yielded three distinct themes relating to recommendations for the campus and recommendations for future African American students. The principal recommendation for the college was very specific: (a) encourage instructors of all backgrounds to build relationships with the African American students they encounter. This particular recommendation from the participants elicits a sense that African American students are a special student population and as such may need additional support mechanisms in place in order to succeed at the rates of other groups. Though the college is a welcoming environment for African American students, this population is still overcoming historical inadequacies in education and preparation for higher education. These reflections also produced two distinct recommendations for future African American students; (a) social integration with the campus culture is critical for success; and (b) developing a personal relationship with your professors is beneficial to your academic and social progress.

All six of the students in this dissertation made reference to the importance of open communication with professors. The comments of the students suggest that the relationships between professors and students must come from a mutual desire to develop partnerships that can enhance the educational experience of the student. Some of the participants viewed the professors on campus as valuable resources that could be mined for information and advice. Additionally professors served as supporters for the participants often challenging them to achieve academically and to broaden their interests. One student remarked that their relationship with their professor during the first year made them want to be at school and knowing that someone other than a family member cared about their academic progress provided a special motivation for success.

The recommendation for the instructors to foster relationships with the students I believe speaks to one of the essential benefits of the community college environment. The participants recognize that not every student will be open to fostering contact with professors outside of the required course content, yet they felt the option would not only appeal to many African American students but would assist them in their transition from high school. Regardless of the professor's ethnic background or academic discipline, each of the participants recognized that having an instructor take an interest in their college career had a significant positive impact of their first year experience.

The recommendations for future African American students included joining student organizations and making their college experience more than attending class and immediately leaving campus for work or home. Each of the participants reflected on the importance of spending time on campus outside of class and the benefits of making social connections on campus that can extend outside of academics. An additional related

recommendation was to emphasize communicating with your professors about any concerns or issues you may be having with the class. One participant reflected that by not asking for help in a math course they were doing themselves a disservice and in the long run not taking full advantage of all of the opportunities afforded to them. Another participant commented on how the extra time spent with a professor during their office hours gave them the confidence to pass an English course they were concerned about failing. Overall the participants stressed that African American freshman should not attempt to take on college alone. They strongly recommend finding and utilizing peers and professors to help with the complex and often confusing world of academia.

Conclusions

The African American students of this dissertation valued the experiences they had at the institution during their first year. Additionally each of the participants reflected that they felt valued and welcomed by the institution thus making it a comfortable environment to further their education. According to the participants the setting provided optimum conditions for participation and for maximizing opportunities for connectedness to the campus. Throughout the interview and reflection process the participants expressed feeling of empowerment from being asked to contribute to research that is designed to help future students. Additionally, the action taken in doing this dissertation created a sense of pride in both the participants and this researcher.

The first and second dialogues with participants generated many reflections on what it means to successfully complete the first year of college. The participants shared through this research their accomplishments and their failures, in an attempt to identify the critical elements that lead to their eventual persistence and academic success.

Students were candid about their interactions with campus programs, services, faculty and staff. Their reflections provided insight into how they felt personally about their first year experience and how they in retrospect view that year.

Students that participated in this dissertation acknowledged a sense of pride in being selected to participate in this research and hoped that their assessment of the first year could be used to help future students avoid any obstacles they may have faced. Additionally they expressed an appreciation in knowing that this dissertation was designed to explore the experiences of African American students specifically, and was created to assist a specific population of students. Each of the participants plan on completing an associates degree from the institution and three of the students are planning on transferring to a four-year university and continuing their education.

These reflections are based on a variety of unique experiences yet yielded some very consistent themes relating to persistence and achievement. Perceptions of the institution were shared and recommendations for future directions were given in an effort to provide insight into a student population in need of assistance. Although this is a small sample of the African American student population, there are some very strong themes that can be extracted from this research and applied to what I consider a population in need of special attention.

Recommendations

Recommendations for Future Research

This dissertation captured the experiences of African American community college sophomores about their experiences as freshman. It's recommended that a mixed method, qualitative study follow this dissertation. An expanded qualitative mixed

methods approach could add additional participants to expand the recommendations collected through analysis of the data combined with statistics of completion and retention rates of the college. Questionnaires could be sent to a larger number of African American students after two consecutive semesters at the college in an effort to include the comments of those who did not continue their education. The comments of those who did not choose to continue could be used to evaluate possible deficiencies the college may be facing.

Additionally a longitudinal study could be implemented to track the progression of African American community college students past the second year. A phenomenological approach to tracking the six participants used in this dissertation over an extended period of time and dialoging with them could serve in further evaluating the student's perceptions of their own progress and failures. Lastly, a comparative study could be conducted between two or more colleges in the same region. This could provide college administrators and instructors an opportunity to learn from what is working or not working from similar institutions and share best practices.

Recommendations for Professional Practice

This dissertation generated specific recommendations for the institution and for future African American students. Based on this dissertation the institution should encourage faculty to make themselves available to students by becoming active as advisors for student clubs or mentoring students that are current majors in the instructor's discipline. These opportunities to interact with students outside of a lecture or lab setting can be extremely valuable to the overall academic development of African American students as they attend college.

The findings indicate that the connections that African American students make with African American professors can be more influential than connections established with non-African American professors. Therefore it is recommended that the institution support a call to action for African American professors to become more involved in the academic progress of the African American students on campus. Additionally it is recommended that African American faculty and staff increase their visibility on campus and utilize the social capital they hold with African American students as a tool for motivation and inspiration.

Furthermore it is recommended that the institution continue to promote and grow its efforts in student services. There is an opportunity for the campus to improve in the areas of open lab hours and open library hours. Participants reflected that the access was vital to their first year academic success and their only concern was the limited hours of operation of these essential programs. The institution should also consider its current strategy for the promotion of student services and evaluate its social networking potential on the World Wide Web.

The research generated recommendations for future African American students.

These recommendations included becoming involved in campus organizations or services early in your academic career. The social support generated through participation clearly assisted the participants in their transition to college and provided them with a support system of other college students to share and compare experiences. Additionally this research suggests that African American students should make and effort to connect with their professors outside of lecture through utilizing office hours or joining student government. The benefits associated with developing a connection to the campus

included a better understanding of course curriculum and opportunities to gain insight into community college culture and practices.

Summary

In summary this research recommends that the college continue its efforts in establishing an environment that simulates the mental and social growth of African American students as they attend college directly out of high school. Through well structured student support services and committed faculty and staff. The institution serves as a gateway for many African American students that are seeking transfer to a four-year institution or professional training for employment. With effective and committed educators and continued opportunities for student support services the institution can continue to promote a culturally competent and intellectually progressive environment for African American students as they attempt to excel in higher education in search of equality and opportunity.

Reflections of the Researcher

My experiences as an African American student and educator have informed my desire to research African American student trends and obstacles in higher education. As a first year college student I earned a cumulative 1.13 grade point average because I was not prepared with the tools or mind set to effectively participate in the higher education experience. The life lessons learned during that first year have informed my approach to learning throughout my undergraduate and graduate education. Most recently that experience has served to provide motivation and inspiration for the development of this research on the African American first year student experience.

My career as an educator has spanned the k-12, community college and university levels. I ventured into the field of education from a legacy of teachers and administrators and knew from early on that I wanted to serve my community and provide inspiration to students of color who were seeking to excel academically and experience college life. I take great pride in my everyday work as a professor and believe that my daily contact with students provides motivation for academic success and a support system for students needing assistance during their college experience.

This dissertation has shown that there are external factors that can contribute to the academic success of African American students and that with support systems in place even students who assess in the lowest percentiles can exceed their own expectations as college students. As a community college faculty member it's my responsibility to my students to make available reasonable tools for success and to do everything in my power to ensure that they are informed of all of the programs and services afforded to them as students of the California community college system.

Knowing that I was neither the first, nor that last African American student to struggle academically during the first semester, this research seeks to create new opportunities for higher education institutions to take a proactive role in the cultivation and support of young African American scholars. We are at a critical junction as a state seeking to increase the percentages of highly trained and highly educated citizens and I believe that an intervention is needed to insure that California's African American students are not overlooked in the efforts to advance these objectives. This research seeks to demystify the African American first year experience and provide insight and understanding to a population in critical need of support.

REFERENCES

- Anonymous. (2008). How black parents perceive urban schools. *The Journal of Blacks in Higher Education*, (60), 36. Retrieved January 4, 2009, from Ethnic NewsWatch (ENW) database. (Document ID: 1555031831).
- Astin, A.W. (1984). Student Involvement: A Developmental Theory for Higher Education. *Journal of College Student Development*, 25(4): 297-308.
- Astin, A.W. (1993). What matters in college: Four critical years revisited. San Francisco: Jossey-Bass.
- Astin, A.W. (1997). How "good" is your institution's retention rate? *Research in Higher Education*, 38(6), 647-658
- Austin, D. (2003). The role of the family influence on student success. In J.A. Ward-Roof & C. hatch (Eds.), *Designing successful transitions: A guide for orientating students to college* (Monograph No. 13, 2nd ed.) (pp.137-163). Columbia, SC: University of South Carolina, national Resource Center for the Freshman Year Experience and Students in Transition.
- Banks, J., & Banks, C. A. M. (Eds.). (2003). *Multicultural education: Issues and perspectives*. New York: Wiley.
- Bell, S.E. (1988). Becoming a political woman: The reconstruction and interpretation of experience through stories. In A.D. Todd & S. Fisher (Eds.), *Gender and discourse*: The power of talk (pp.97-123). Norwood, NJ: Ablex
- Berger, J.B. & Milem, J.F. (1999). The role of student involvement and perceptions of integration in a causal model of student persistence. *Research in Higher Education*, 40 (6), 641-664.
- Bennett, L. (1993). *Before the Mayflower: A history of Black America* (6th ed). New York: Penguin
- Blackwell, J.E. (1983). Strategies for improving the status of Blacks in higher education. Planning and Change, 14(1), 56-73
- Bryant, M., (2006). Talking about change: Understanding employee responses through qualitative research. *Management Decision*, 44(2), 246-258. Retrieved May 4, 2007, from ABI/INFORM Global database. (Document ID: 1032563941).
- Cawthorn, T.W., & Miller, M. (2003). *Today's students and their impact on orientation and first year programs*. In J.A. Ward-Roof & C. hatch (Eds.), Designing successful transitions: A guide for orientating students to college (Monograph

- No. 13 2nd ed.) (pp. 1-13). Columbia, SC University of South Carolina. National Resource Center for the Freshman Year Experience and Students ii Transition.
- California Community Colleges System Office. (2007). Accountability reporting for the California Community Colleges' (A report to the Legislature Pursuant to AB 1417). Sacramento, CA: Technology Research and Information Systems.
- California Department of Education High School Exit Exam Office. (2008).

 California High School Exit Exam Results for Mathematics and English language
 Arts by Gender and Ethnic Designation State Report. Retrieved January 21, 2009,
 from http://dq.cde.ca.gov/dataquest/cahsee/ExitEth1.asp?cLevel
 =State&cYear=2007-08&cChoice=ExitEth1&cAdmin=C&tDate=000000&
 TestType=E&cGrade=10&Pageno=1
- California Community Colleges Chancellor's Office (CCCCO) (2007). Focus on Results: Accountability Reporting for California Community Colleges (ARCC). http://www.cccco.edu/divisions/tris/rp/ab_1417/ARCC_Report_2007.pdf
- Community College League of California. (2008). Fast Facts, Sacramento, CA.
- Cosumnes River College Research Office Report. (2007). Newsletter, Sacramento, CA
- CRC. (2009). *About CRC/CRC History*. Retrieved April 13, 2009, from http://www.crc.losrios.edu/About_CRC/CRC_History.htm
- Crenshaw, K., Gotanada, N., Peller, G., and Thomas, K.(eds.) (1995). *CriticalRace Theory: The Key Writings that Formed the Movement*. New York: Free Press.
- Creswell, J.W. (2005). *Educational research: planning, conducting, and evaluating quantitative and qualitative research.* (2nd ed.). New Jersey: Pearson Education.
- Crosson, P.H. (1988). Four-year college and university environments for minority degree achievement. Review of Higher Education, 11, 205-382
- Daniel, G. S. and O. Armida (2004). A critical race analysis of Latina/o and African American Advanced Placement enrollment in public high schools. *The High School Journal* 87(3): 15.
- Delgado, R., & Stefancic, J. (2001). *Critical Race Theory*. New York: New York University Press.
- Delgado, R., (1995). *Critical race theory: The cutting edge*. Philadelphia, PA: Temple University Press.

- Dowd, A.C. (2007). Community Colleges as Gateways and Gatekeepers: Moving beyond the Access "Saga" Toward Outcome Equity. *Harvard Educational Review*, (77), p.407
- Durkheim, E., Simpson, G., Spaulding, J.A. (1951). *Suicide*. New York, NY: The Free Press.
- Elliott, J. (2005). *Using narrative in social research: Qualitative and quantitative approaches.* Thousand Oaks, CA: Sage Publications.
- Ethington, C.A. (1990). A psychological model of student persistence. *Research in Higher Education*, 31 (3), 279-293
- Ford, B., Obiakor, F., Patton, J.(eds.) (1995). Effective education of african american exceptional learners: New Perspectives. Austin, TX: Pro-Ed
- Gardner, J.N., & Hansen, D.A. (1993). Perspective on the future of orientation. In M.L. Upcraft (Ed.), *Designing successful transitions: A guide for orientating students to college* (Monograph No. 13, pp.183-194) Columbia, SC: National Resource Center for the Freshman Year Experience, University of South Carolina.
- Gates, E.N. (1997). *The concept of race in natural and social science*. Florence, KY: Routledge.
- Gordon, E. W. (1999). *Education and justice: A view from the back of the bus*. New York, NY: Teachers College Press.
- Green, M.F. (1989). *Minorities on campus: A handbook for enhancing diversity*. Washington, DC: The American Council on Education.
- Grosset, J.M. (1991). Patterns of integration, commitment, and student characteristics and retention among younger and older students. *Research in Higher Education*, 32 (2), 159-178
- Hale, J. (2001). Learning while Black: Creating educational excellence for African American children. Baltimore, MD: John Hopkins University press.
- Harper, S.R. (2006). Responding to the crisis concerning African American men in college: Implications for practice and policy. Washington, DC: Joint Center for Political and Economic Studies, Dellums Commission.
- Harper, S.R., & Wolley, M.A. (2002). Becoming an "involving college" for African American undergraduate men: Strategies for increasing African American male participation in campus activities. *The Bulletin of the Association of College Unions International*, 70(3), 16-24.

- Hefner, D. (2004). Where the boys aren't. *Black Issues in Higher Education*, 21(9), 70-75.Retrieved April 26, 2007, from Research Library database. (Document ID: 659674191).
- Hirsch, E.D. (2006). *The knowledge deficit: Closing the shocking education gap for American children*. Boston: Houghton Mifflin Company.
- Jalomo, R. (1995). Latino students in transition: An analysis of the first-year Experience in Community College. Unpublished doctoral dissertation, Arizona State University
- Johnson, D.B. (2002). Student persistence: A study of student pre-enrollment factors and freshman to sophomore persistence at a public university. Unpublished doctoral dissertation, the University of Alabama.
- Kuh, G. D. (2003). What We're Learning About Student Engagement From NSSE: Benchmarks for Effective Educational Practices. *Change*, *35*(2): 24-32.
- Kuh, G. D., Schuh, J. H., Whitt, E. J., Andreas, R., Lyons, J., Strange, C. C., Krehbiel, L. E., and MacKay, K. A. (1991). *Involving Colleges: Successful approaches to fostering student learning and development outside the classroom*. San Francisco: Jossey-Bass Publishers.
- Levitz, R., & Noel, L. (1989). Connecting students to institutions: Keys to retention and Success. In M.L. Upcraft & J.N. Gardner (Eds.), *The freshman year experience: Helping students survive and succeed in college.* (pp. 65-81). San Francisco: Jossey-Bass
- Los Rios Office of Human Resources (2006). Regular Faculty Demographics Report.
- Ottinger, C. (1991). College-going persistence and completion patterns in higher education: What do we know? *Research Briefs*. 2(3). Division of Policy Analysis and Research Washington, DC: American Council on Education
- Perkins, B. (2008). What We Think: Parental Perceptions of Urban School Climate Alexandria, VA: National School Boards Association.
- Rendon, L. I., Jalomo, R. E., and Nora, A. (2000). Theoretical consideration in the study of minority student retention in higher education. In *Rethinking the Departure Puzzle: New Theory and Research on College Student Retention*, edited by J. Braxton, 127-156. Nashville, TN: Vanderbilt University Press.
- Riessman, C. K., (1993). Narrative analysis. Newbury Park, CA: Sage Publications.
- Roberts, T.W. (1995). Differentiation from the family of origin and adjustments to college in young adults. *Journal of the Freshman Year Experience*, 7(1),

- Shneiderman, B. (2003). Leonardo's Laptop. Cambridge, MA: The MIT Press.
- Spady, W. (1970). Dropouts from higher education: An interdisciplinary review and synthesis. *Interchange*. 1, 64-85
- Stovall, D. (2004). "SCHOOL LEADER AS NEGOTIATOR: Critical Race Theory, Praxis, and the Creation of Productive Space." *Multicultural Education* 12(2): 8.
- Strumpf, G., & Sharer, G. (1993). Trends and issues in orientation programs. In M.L. Upcraft (Ed.), *Designing successful transitions: A guide for orientating students to college* (Monograph No. 13, pp.27-34) Columbia, SC: National Resource Center for the Freshman Year Experience, University of South Carolina.
- Tatum, B.D. (2007). Can we talk about race? and other conversations in an era of school resegregation. Boston: Beacon
- Tinto, V. (1975). Dropout from higher education: A theoretical synthesis of recent research. *Review of Educational Research*, 45, 89-125.
- Tinto, V. (1993). Leaving college: *Rethinking the causes and cures of student attrition* (2ndEd.). Chicago: The University of Chicago Press.
- U.S. Census Bureau, (2000). *State and county quick facts*. Retrieved April 14, 2009, from http://quickfacts.census.gov/qfd/meta/long_68176.htm
- Van Gennep, A. (1960). *The rites of passage*. Translated by M. Vizedon and G. Caffee. Chicago: University of Chicago Press. Originally published as *Les rites de passage*. Paris: Nourry, 1909.
- Ward-roof, J.A. & hatch, C. (2003). *Designing successful transitions: A guide for orienting students to college, 2nd edition* (Monograph No. 13). Columbia, SC: National resource Center for the Freshman Year Experience, University of South Carolina.
- West, C. (1992, August.2). Learning to talk of ace. New York Times Magazine, 24, 26.
- Williams, P. (1991). Alchemy of race and rights. Cambridge: Harvard University Press.
- Wolcott, G.V. (2006). Critical reflections of college sophomores about preparation and persistence during their freshman year. Unpublished doctoral dissertation, the University of San Francisco.

- Woodson, C. (1933). *The mis-education of the negro*. Nashville, TN: Winston-Derek Publishers.
- Young, C. Y., J. V. Wright, et al. (2005). Instructing African American Students. *Education; Chula Vista 125*(3): 516-525.

APPENDIX A

IRBPHS Initial Application

APPENDIX A

IRBPHS Initial Application

IRBPHS INITIAL APPLICATION

Name of Applicant: Robert Snowden Jr

USF Identification Number: 10163048

University Title: The University of San Francisco

School or College: School of Education

Department or Group: International and Multicultural Education

Home or Campus Address:

5616 Birdview Way Elk grove, CA 95757

Home Phone: 916-548-2512 **Work Phone:** 916-691-7666

Electronic Mail Address(s): snowden76@gmail.com, rlsnowden@dons.usfca.edu

Name(s) and University Title(s) of Other Investigators:

Name of Faculty Advisor: Dr. Betty Taylor

University Title: Professor

Home or Campus Address:

School of Education

Department of International and Multicultural Education

Office: SOE, Room 220 San Francisco, CA 94117

Home or Campus Phone: (415) 422-6041

Electronic Mail Address(s): taylorb@usfca.edu

Signature of Applicant Date

Signature of Faculty Advisor* Date

significant problems or changes.

Project Title: CRITICAL REFLECTIONS OF AFRICAN-AMERICAN CALIFORNIA COMMUNITY COLLEGE STUDENTS ABOUT FACTORS LEADING TO SCHOLASTIC SUCCESS DURING THEIR FIRST YEAR

Respond to items 1 - 11 white paper, single-sided, typed in black ink and using standard 12 point font.

Responses to items 1-11 should be stapled to this Initial Application form.

African American students are facing challenges in matriculating from high school in to institutions of higher education within the state of California. Systematically, the high percentage of African American students that are educated in California's low or underperforming schools leads to a direct correlation to the dismal numbers of African American students that arrive adequately prepared for the academic rigors of higher education. The issues with education are vast and without intervention students of color will continue to fail in successfully navigating the requirements of colleges and universities. Within the state of California an increasing diverse population is eventually going to force institutions in of higher learning to address the problems associated historically underperforming populations.

The California Community College (CCC) system currently serves the most diverse population in its history. Unfortunately African American students attending the community colleges of the state are facing overwhelming social and academic challenges and are consequently failing to achieve proportionate graduation and transfers rates compared to other populations statewide. There are a variety of challenges facing institutions that both in theory and practice seek to establish equity between marginalized populations without limiting the opportunities for other groups. As the CCC system continues to grow steps must be taken to institutionalize methods that consistently address issues associated with multicultural education. Despite the fact the CCC system has traditionally been a valued part of the collegial matriculation process for millions of students, yet with fewer and fewer African Americans successfully navigating the system current educational trends will undoubtedly continue.

To date a noteworthy amount of research has been conducted on African American student success strategies and the first year experience of college students independently. What remains to be explored are the critical reflections and experiences of first year African American CCC students about their persistence, progress towards degree/certificate completion, transfer preparation and their use of student services resources. This work is extremely relevant considering the current trends in education that are perpetuating educational inequality among students of color.

^{*}Your signature indicates that you accept responsibility for the research described, including work by students under your supervision.

It further attests that you are fully aware of all procedures to be followed, will monitor the research, and will notify the IRPBHS of any

The purpose of this study is to explore through the reflections of students the factors that lead to African American community college freshman persistence. This study will: 1) examine college programs or services that were available to or utilized by the student during the first year and, 2) explore the recommendations of African American college sophomores for incoming African American first semester students. For the purpose of this study student persistence will be defined as continued attendance and unit completion from the semester of first enrollment through two consecutive semesters as a full time student according to number of units attempted and completed. Limiting the study to a northern California urban/suburban community college expands upon previous research on freshman success, African American college student success, and African American students in the California community college system.

This study will examine the experiences of African American students at Cosumnes River College in Sacramento California who have completed their first year of academic courses. The significance of this study in embedded within a focus on a segment of the community college student body that has been identified by the college as suffering from poor matriculation and persistence rates. This researcher's intention is that the information gathered from the students' reflections, can lead to the overall enhancement of efforts to retain and support African American college freshmen. This research will serve the campus by informing administrators on student perspectives in relation to their own education and subsequently motivating administrators and faculty to consider reasonable improvements to the current college infrastructure.

Though community colleges can serve as gateways to higher education, they can also work to the benefit of four-year institutions as gatekeepers (Dowd, 2007). As a gatekeeper many undergraduate students seeking transfer are locked into completing their entire lower division course work at the community college before they can be admitted to the four-year of their choice. Despite a growing number of students enrolling in community colleges and the systems ability to assist in transfer to four-year institutions, the current organization of community colleges is providing both a democratization and diversion effect on its student population. Current research supports the concept that the United States is facing a crisis in regards to the scholastic preparedness of African American students (Hefner, 2004). African American students specifically in the state of California unfortunately continue to struggle in achieving their transfer goals and improving their rates of matriculation. Therefore it becomes critical that community colleges take a proactive role in supporting this marginalized population in the attainment of their educational goals.

2. Description of Sample

This researcher proposes to invite six African American students to share reflections on their experiences as first year community college students and serve as active participants in the improvement of institutional programs and services. Subjects selected for this study will have met the following criteria: (1) Currently enrolled as a 2nd year student at Cosumnes River College (2) Will have completed at least 24 units of academic

coursework at Cosumnes River College by the Fall 2008semester (3) Are in good academic standing:

- a. Subjects will be current students at Cosumnes River College.
- b. The selected participants have been at their respective school at least one year and have successfully matriculated to a second year student status in good academic standing.
- c. Applicant has obtained access to subjects through employment in the Los Rios Community College District
- d. Subjects will not be interviewed as part of any course or some other type of captive capacity.

3. Recruitment Procedure

- a. Faculty members at Cosumnes River College who teach intermediate or second year curriculum will be asked by the applicant through an email (Appendix F) to disseminate a letter (Appendix E) to their African American students. The letter introduces the project and states that if the students fit the described selection criteria and if they are willing to participate that they are invited to contact the researcher in order to set up an informational meeting. This recruitment process has been proposed so that the researcher can deliberately gather representation of the African American student population from across multiple academic disciplines.
- b. An email will invite self-selected participants to an informational session at an off campus location to further explain the study. At that time potential participants will be offered a consent form and a questionnaire designed to gather information about their first year experience.

4. Subject Consent Process

- a. No parental consent is necessary; the subject is over 18 years of age.
- b. The project will involve two face-to-face interviews between the applicant and the subjects.
- c. The project will include a short invitation letter attached (Appendix E)
- d. Project does not involve simultaneous administration of a survey or questionnaire to a large group of subjects.
- e. Not applicable.
- f. Consent form that will be used is from the University of San Francisco's IRBPHS Appendix B.

5. Procedures

a. Subject's transcripts will be observed to verify that they are current students at Cosumnes River College and that they have completed two consecutive semesters between Fall 2007 and Fall 2008

- b. Subjects will be provided a short letter, questionnaire and informed consent form.
- c. Subjects will return a signed copy of consent form.
- d. Subjects will be sent up a time for the first and second interview to be held at a local off campus location.
- e. Questionnaire will be collected on the day of the interview.
- f. Subjects will participate in 2 face-to-face interviews.
- g. Interviews will be transcribed by applicant after each session.
- h. Transcriptions will be sent to subjects to review for accuracy.

6. Potential Risks to Subjects

There may be some emotional discomfort as participants share some of their personal and educational experiences. Subjects may also be concerned about the confidentiality of their identities when asked about their experiences as students.

7. Minimization of Potential Risk

In an effort to provide protection to the subjects, participants will select pseudonyms for themselves to be identified by in interviews and in the reporting of the data. Moreover the researcher will apply a pseudonym for every instructor, manager or staff member indentified in the interview sessions. Lastly the institution will not be directly identified and by holding the interviews in a local off campus location the subjects are further assured an anonymous experience.

The researcher will attempt to minimize stress or discomfort to the participant by making the interview environment a conference or meeting room off campus but within close proximity to the college. Additionally the questionnaire will be coded for record keeping purposes and remain confidential. The applicant has chosen to conduct two interview sessions. The first one will obtain information and the second will be used for clarification of information obtained in the first session.

8. Potential Benefits to Subjects

Subjects may gain personal self-awareness and empowerment knowing that they were coresearchers on a body of knowledge that will be available for others to read and learn from. The anticipated result is to inform best practices of California Community Colleges in the implementation of programs and courses designed to assist first time freshman in attaining their academic goals. The academic community can benefit from this research by applying its findings in the development of curriculum that prepares first time freshman with the tools necessary for scholastic achievement and matriculation through higher education.

9. Costs to Subjects

There are no anticipated costs to the participants of this study other than the time provided for the interviews. Each interview will be approximately one-half hour to one hour in length. There may also be additional time required of subjects to read and verify their transcripts. The researcher will encourage participants to meet in a place that is convenient to them.

10. Reimbursements/Compensation to Subjects

No reimbursements will be involved in the procedures, as there will be no monetary cost to the participants of this study. Additionally, no compensation will be given to participants. Participants will be made aware of these stipulations through consent.

11. Confidentiality of Records

The transcripts and audio tape of the interviews from this project will be kept under lock and key at Cosumnes River College Room L-101, 8401 Center Parkway, Sacramento, CA 95823 for approximately 1 year after the project has been completed then the material will be shredded and destroyed.

APPENDIX B

Los Rios Community College District research Approval Letter

Final: 9 20-05

LOS RIOS COMMUNITY COLLEGE DISTRICT REQUEST TO CONDUCT RESEARCH ABOUT LRCCD STUDENTS AND/OR COLLEGES

Research Proposal Cover Sheet

ress: 5616 Birdview Way ress: snowden76@gmail.com uployees: College: Cosumnes rad Researcher: College/Univers Address: Program of Stud	American Community College Stud	Work Phone: (916) 691-7666
ress: 5616 Birdview Way ress: snowden76@gmail.com uployees: College: Cosumnes rad Researcher: College/Univers Address: Program of Stud	American Community College Stud	Home Phone: (916) 548-2512 Work Phone: (916) 691-7666 Phone:
ress: 5616 Birdview Way ress: snowden76@gmail.com uployees: College: Cosumnes rad Researcher: College/Univers Address: Program of Stud	y/Organization:Fa	Home Phone: (916) 548-2512 Work Phone: (916) 691-7666 Phone:
ress: snowden76@gmail.com uplayees: College: Cosumnes uad Researcher: College:Univers Address: Program of Stud Group Responsible for Study Ove	y/Organization:Fa	Work Phone: (916) 691-7666 Phone:
unlayees: College: Cosumnes and Researcher: College/Univers Address: Program of Stud Group Responsible for Study Ove	y/Organization:Fa	Work Phone: (916) 691-7666 Phone:
and Researcher: College/Univers Address: Program of Stud	y/Organization:Fa	Phone:
Address: Program of Stud	: Fa	Phone:
Program of Stud	Fa	aculty advisor:
	sight Approval (if appropriate):	
vestigator's Faculty Advisor OR		
	Department Chair OR Professional	Standards Committee:
	Signature:	
	Date:	
for LRCCD College or District F	esources (use of space, use of colle	ege data, etc.):
t is a Modification or Renewal, do		
Following:		
tion; subjects required (number, sou kground and experience of investig	ce/recruitment procedures, criteria fo	or inclusions/exclusions, etc.); name(s, academic
y of university agency research app	oval, including approved guidelines	for human subjects in the research (where appropriate
sent form for participants (if conser	from participants is needed to condu	uct the research)
ft survey instrument and/or intervie	protocol	
i i	s of additional investigators: s for LRCCD College or District Resident of the proposed study: Modification it is a Modification or Renewal, does No e Following: research proposal abstract that defineration; subjects required (number, sour eleground and experience of investigately of survey instrument (where appropriately of survey instrument) and or interviewed forms for participants (if consentations of the proposal abstract that definerations are proposal abst	s of additional investigators: s for LRCCD College or District Resources (use of space, use of college in this proposed study: Modification Renewal study: it is a Modification or Renewal, does this submission differ in any was segment to the submission of the submission differences in the submission di

APPENDIX C

Student Invitation Letter

APPENDIX C Student Invitation Letter

September 30, 2008

Dear Cosumnes River College Student,

My name is Robert Snowden and I am a doctoral student in the School of Education at the University of San Francisco. I am conducting a study on the experiences of second year African American students at Cosumnes River College (CRC). You were selected to receive this invitation because college faculty members recommend you based on their knowledge of your academic history. Please consider participating in this study if you meet the following criteria: You are a 18-20 years old, an African American student at CRC, this is your second consecutive year at CRC, you have completed at least 24 units, you are currently a fulltime student and lastly that you are in good academic standing.

If you agree to participate in this study, you will be asked to provide an unofficial copy of your transcript and you will be interviewed on two separate occasions, at a place and time convenient for you. If you were to feel uncomfortable in answering any of the questions or performing any of the tasks within this study, you would be free to decline or stop at any time. All study information will be kept in locked files at all times. Only study personnel will have access to the files. No individual identities will be used in any reports or publications resulting from this study.

The anticipated benefit of this work will be in informing best practices of California Community Colleges in the implementation of programs and courses designed to assist first time freshman in attaining their academic goals. There would be no cost to you as a result of taking part in this study, nor would you be reimbursed for your participation in this study. Lastly, if you meet the criteria and are not selected to participate in the study thank you in advance for your time and energy.

If you have questions about the research, or if you believe that you would like to participate you may contact me at snowden76@gmail.com or at (916) 548-2512. If you have further questions about the study, you may contact the IRBPHS at the University of San Francisco, which is concerned with the protection of volunteers in research projects. You may reach the IRBPHS office by phoning (415) 422-6091 and leaving a voicemail message, by e-mailing IRBPHS@usfca.edu, or by writing to the IRBPHS, Department of Psychology, University of San Francisco, 2130 Fulton Street, San Francisco, CA 94117-1080.

Sincerely,

Robert Snowden Doctoral Student University of San Francisco

APPENDIX D

Letter to College Faculty

APPENDIX D

Draft Letter to College Faculty

September 30, 2008

Dear Colleague:

As you know I am a doctoral student in the School of Education at the University of San Francisco. I am conducting a study on the experiences of second year African American students at Cosumnes River College (CRC). I am requesting your assistance in helping me generate a sample of students that meet specific criteria for participation in this study. If you would as by disseminating the attached letter (Appendix E) to students that you identify as second year African American's between the ages of 18-20. If the identified student agrees to participate in the study the letter instructs them on how to contact me via email at snowden76@gmail.com or by phone at 916-548-2512.

Thank you for you assistance,

Robert Snowden Doctoral Student University of San Francisco

APPENDIX E

Release Form for Tapes and Transcripts

APPENDIX E

Release Form for Tapes and Transcripts

I agree to have my dialogues with Robert Snowden audio taped and transcribed into written form. I realize that it is the responsibility of both Mr. Snowden and myself to ensure the accuracy of these transcriptions.

I am also aware that the original audio tapes and transcripts will be kept safe and secure in the office of Mr. Snowden at Cosumnes River College at 8401 Center Parkway, Sacramento, CA 95823 for period of one year, after which they will be destroyed. The transcribed copies I receive will be destroyed or not at my discretion.

Subject Signature	Date	
Signature of Person Obtaining Consent	Date	

APPENDIX F

Interview Protocol – Dialogue Questions

APPENDIX F

Interview Protocol – Dialogue Questions
The qualitative study proposes to inquire into three primary questions:

- 1. What factors led to the persistence of African American students during their first year at a California community college?
- 2. Have college generated programs or services impacted the success of African American students during the first year at a California community college?
- 3. What recommendations do current African American students have for the institution and future African American first year California community college students?

The areas of interest for the interview protocol questions is as follows:

- 1. Please describe yourself, where are you from what are your interests and goals.
- a. Please describe how you came to the decision to attend community college, and what where the factors that led to your choice in Cosumnes River College?
- b. What were some of your success and failures during your first year?
- c. Please describe your experiences with college faculty, administrators or staff during your first year.
- d. Please describe your first semester as a student at Cosumnes River College.
- 2. Please describe any opportunities you had to utilize campus services or facilities.
- a. What were your expectations from the college in regards to services provided and your overall student experience?

- b. Please describe your experiences with using technology for school related activities during your first year.
- c. Please describe your perceptions on the availably of campus services specifically for the social or academic support of African American students at Cosumnes River College.
- d. Please describe your social interactions while on campus during your first year at Cosumnes River College.
- e. Please describe your overall academic performance during your first year at Cosumnes River College.
- 3. Please describe what you feel the institution and future freshman could learn from your experiences as a first year community college student.
- a. What helped you successfully complete your full first year at Cosumnes River College?
- b. As an African American student did you at any point during your freshman year feel excluded or not apart of the campus culture?
- c. What recommendations would you give to the institution to further assist African American students in achieving their academic goals?

APPENDIX G

Informed Consent Letter

Appendix B

INFORMED CONSENT FORM

UNIVERSITY OF SAN FRANCISCO CONSENT TO BE A RESEARCH SUBJECT

Purpose and Background

1. Robert Snowden, a graduate student in the School of Education at the University of San Francisco is conducting a study on the first year experience of African American students at a California Community College

I am being asked to participate because I am second year African American student between the ages of 18-20 at a California Community College

If I agree to be a participant in this study, the following will happen:

- 1) I will complete a short questionnaire my first year experience.
- 2) I will participate in an interview with the researcher, during which time I will be asked about my experiences as a first time freshman.

I will complete the questionnaire and participate in the interview at a place of my own convenience.

Risks and/or Discomforts

- 1. I am free to decline to answer any questions that I do not wish to answer or that make me feel uncomfortable.
- 2. Confidentiality will be maintained as I participate in the study. Records will be kept confidential and no individual identities will be used in any reports or publications resulting from this study. Study information will be kept in locked files at all times.

Benefits

There will be no direct benefit to me from participating in this study. The anticipated benefit of this study is a better understanding of the effects of leadership styles and behaviors on student academic performance.

Cost/Financial Considerations

There will be no financial costs to me as a result of taking part in this study.

Payment/Reimbursement

I will not be paid for my participation in this study.

Questions

I have talked Robert Snowden about this study and have had my questions answered. If I have further questions about the study, I may contact him at (916) 691-7666.

If I have any questions or comments about participation in this study, I should first talk to the researcher. If for some reason I do not wish to do this, I may contact the IRBPHS which is concerned with the protection of volunteers in research projects. I may reach the IRBPHS office by calling (415) 422-6091 and leaving a voicemail message, by e-mailing IRBPHS@usfca.edu, or by writing to the IRBPHS, Department of Psychology, University of San Francisco, 2130 Fulton Street, San Francisco, CA 94117-1080.

Consent

I have been given a copy of the "Research Subject's Bill of Rights" and I have been given a copy of this consent form to keep.

PARTICIPATION IN RESEARCH IS VOLUNTARY. I am free to decline to be in this study, or to withdraw from it at any point. My decision as to whether or not to participate in this study will have no influence on my present or future status as an employee with my employer. My signature below indicates that I agree to participate in this study.

Subject's Signature	Date of Signature	
	<u>——</u>	
Signature of Person Obtaining Consent	Date of Signature	

APPENDIX H

Demographic Questionnaire

Demographic Questionnaire

Name		
Age	_	
Sex		

- 1. Current Major:
- 2. Please list your cumulative G.P.A after your first year at the institution:
- 3. Did you have a job during your freshman year? If so where and how many hours did you work?
- 4. During your first year were you involved in any campus activities or organizations? If so please list.
- 5. Name of High School you graduated from. City and State
- 6. Please list your career goals and objectives:
- 7. Please state the highest education level of your mother and father:
- 8. Please list any additional information you would like to share about your self or your experiences at the college: